

iaw@iaw.edu | www.iaw.edu

IAW Diskussionspapiere

Nr. 138 Juli 2022 | No. 138 July 2022

Parenting Styles, Socioeconomic Status and (Non-)Cognitive Skills

Philipp Kugler Martin Kroczek Anne Zühlke

Institut für Angewandte Wirtschaftsforschung e.V. Schaffhausenstraße 73 | 72072 Tübingen | Germany Tel.: +49 7071 98960 | Fax: +49 7071 989699

ISSN: 1617-5654

IAW-Diskussionspapiere

Das Institut für Angewandte Wirtschaftsforschung (IAW) Tübingen ist ein unabhängiges außeruniversitäres Forschungsinstitut, das am 17. Juli 1957 auf Initiative von Professor Dr. Hans Peter gegründet wurde. Es hat die Aufgabe, Forschungsergebnisse aus dem Gebiet der Wirtschafts- und Sozialwissenschaften auf Fragen der Wirtschaft anzuwenden. Die Tätigkeit des Instituts konzentriert sich auf empirische Wirtschaftsforschung und Politikberatung.

Dieses IAW-Diskussionspapier können Sie auch von unserer IAW-Website als pdf-Datei herunterladen:

https://www.iaw.edu/iaw-diskussionspapiere.html

ISSN 1617-5654

Weitere Publikationen des IAW:

- IAW-News (erscheinen 4x jährlich)
- IAW-Impulse
- IAW-Kurzberichte
- IAW Policy Reports
- IAW-Forschungsberichte

Möchten Sie regelmäßig eine unserer Publikationen erhalten, dann wenden Sie sich bitte an:

IAW Tübingen e.V. Schaffhausenstraße 73, 72072 Tübingen Telefon: 07071 98960 Fax: 07071 989699 iaw@iaw.edu

Aktuelle Informationen finden Sie auch im Internet unter:

https://www.iaw.edu/

Der Inhalt der Beiträge in den IAW-Diskussionspapieren liegt in alleiniger Verantwortung der Autorinnen und Autoren und stellt nicht notwendigerweise die Meinung des IAW dar.

Parenting Styles, Socioeconomic Status and (Non-)Cognitive Skills

Philipp Kugler * Ma

Martin Kroczek †

Anne Zühlke[‡]

July 22, 2022

Abstract: This paper analyzes the role of parenting styles, a recent topic in the economic literature. Using a novel latent class model, we investigate which parenting styles can be observed in the data and how parenting styles are related to parents' socioeconomic status and household composition. We identify four parenting styles. An authoritarian and an authoritative style closely resemble the styles proposed by psychologists. The two other styles are variations of these styles. The parenting styles are strongly associated with household income, education and whether a child is an only child. The results suggest that constraints in both time and (non-)cognitive skills of the parents restrict their choice. We find that children's skills, in particular non-cognitive skills, are strongly associated with the parenting style. Parenting styles that are associated with low household income and having more than one child are associated with lower skills of the child. Therefore, our results indicate that parenting styles might be an important factor in explaining the skill gap in early childhood between children from different socioeconomic origins.

Keywords: human capital, skills, parenting, child rearing, parenting style, social mobility, socio-economic status, topic modeling JEL-Codes: J13, J24, J62

Declarations of interests: The authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

^{*}Corresponding author: Philipp Kugler, Institute for Applied Economic Research, Schaffhausenstr. 73, 72072 Tübingen, Germany; Email: philipp.kugler@iaw.edu.

[†]Martin Kroczek, Institute for Applied Economic Research, Schaffhausenstr. 73, 72072 Tübingen, Germany; Email: martin.kroczek@iaw.edu.

[‡]Anne Zühlke, Institute for Applied Economic Research, Schaffhausenstr. 73, 72072 Tübingen, Germany; Email: anne.zuehlke@iaw.edu.

^{‡‡}We would like to thank Martin Biewen for valuable discussions and comments. This paper uses data from the National Educational Panel Study (NEPS; see Blossfeld & Roßbach 2019). The NEPS is carried out by the Leibniz Institute for Educational Trajectories (LIfBi, Germany) in cooperation with a nationwide network.

1 Introduction

The ambition to provide equal chances for economic and social participation to every child is broadly voiced among developed societies. Yet, a large body of literature documents a gap in non-cognitive and cognitive skills across parental income and education, even in early childhood (see Heckman & Mosso 2014, Francesconi & Heckman 2016, Attanasio 2015, for extensive reviews of the recent literature). At the same time, early childhood factors are important determinants of economic and social adult outcomes. For example, Keane & Wolpin (1997), Cunha et al. (2005) and Huggett et al. (2011) show that at least half of the lifetime income variability across individuals arises from differences in childhood characteristics, which are primarily influenced by children's environment. In this sense, Cunha et al. (2006) underline the importance of parenting, i.e. all actions taken by parents to support the development of their child. Parenting has been the subject of economic research for several decades, dating back at least to the work on families by Becker (1981) and Becker & Tomes (1986).

Various studies that analyze the behavior of parents, focus on parental investment. In particular, they show how time and monetary investments affect children's skill acquisition and how such investments differ between parents with different socioeconomic status (e.g Cunha et al. 2013, Boneva & Rauh 2016, Attanasio et al. 2020, Falk et al. 2021). Parenting styles, the broad strategy of how parents interact with their children, can be seen as another dimension of investment. However, the choice and effects of parenting styles is a rather novel topic in the economic literature (see Doepke et al. 2019, for an extensive review of the economic literature on parenting).

We supplement and extent the research on parenting styles. Our data contains a large set of measures about the parent-child interaction from six different domains, such as how parents monitor their child or how much autonomy parents leave to their child. Using Latent Dirichlet Analysis for Survey Data (LDA-S), a hierarchical model recently proposed by Munro & Ng (2022), we operationalize parenting styles as latent classes. LDA-S differs from conventional models used to recover latent factors. First, LDA-S acknowledges that the survey responses on parent-child interaction are categorical. Second, it provides an economic interpretation to the unobserved heterogeneity. Finally, LDA-S connects unobserved heterogeneity with observed characteristics and survey responses. Therefore, we can directly incorporate and analyze differences in parenting styles along parental socioeconomic status and household composition. Further, we analyze how parenting styles are associated with children's non-cognitive and cognitive skills.

Our paper connects at least three strands of literature. First, we add to the discussion on how to operationalize parenting styles. Economists analyzing parenting styles commonly refer to the theoretical foundations laid out by Baumrind (1971, 1991). The framework classifies parents' behavior into two dimensions, demandingness and responsiveness. This results in four different parenting styles. *Authoritative* parents are both demanding and responsive. This style is defined by parents monitoring and communicating clear rules and standards for their children's behavior. Parents are assertive, but not invasive or restrictive. They support their children rather than punish them with disciplinary methods and raise them to be socially responsible, self-regulated and cooperative. *Authoritarian* parents are demanding as well, but not responsive. This parenting style is characterized by an orderly environment without explanations and a clear set of regulations. These parents are obedience- and status-

oriented. In contrast, *permissive* parents are more responsive and less demanding. They allow self-regulation and avoid confrontation. This parenting style is non-traditional and tolerant. Last, *neglecting* parents are neither demanding nor responsive. This style is defined by non-supportive parents, who do not monitor their children, but actively reject them.

Doepke & Zilibotti (2017) develop a theory that rationalizes the choice between Baumrind's parenting styles. The equilibrium of the model results in different parenting styles depending on parental preferences and the socioeconomic environment. Other theoretical models capture key features of Baumrind's parenting styles. For example, Burton et al. (2002) model parenting styles as parent's patience when the child misbehaves. Lundberg et al. (2009) model parenting styles as the control of the parents on the child's decision-making. The model proposed by Cobb-Clark et al. (2019) captures the closeness of the parent-child relationship and the degree of monitoring parents employ. In empirical studies, the parenting styles are commonly modeled as continuous latent factors coming from factor analysis. Cobb-Clark et al. (2019) get two indices to measure parenting styles. One capturing whether parents respect the child's views and opinions, the other how much the parents monitor the child. Falk et al. (2021) rely on three domains of parent-child interaction (i) parental warmth, (ii) parental interest and monitoring, and (iii) parental psychological and behavioral control. They recover one latent factor, for which higher values reflect warm and child-oriented parenting but also a high degree of monitoring, while a lower value is associated with a higher degree of punishment. Fiorini & Keane (2014) identify two latent factors. One is an index of warmth and affection, the other can be interpreted as the effectiveness of imposing discipline.

The second major literature we connect to is that on the relation between parental investment and socioeconomic status. Recent literature established a strong link between both and discusses potential causal channels. Parental time and monetary investment may hinge on parents' objective, resource constraints and incorrect beliefs about the child's production function of human capital (Attanasio 2015, Dizon-Ross 2016, Doepke & Zilibotti 2017). Evidence on the link between parenting styles and socioeconomic status is scarce. In their theoretical framework, Cobb-Clark et al. (2019) model parenting styles as parental investments. The investment depends not only on time and income but on mental effort required to pay attention to engage with, monitor and supervise the child. Their model links socioeconomic status to parental investment by allowing the endowment of a household's attention to depend on socioeconomic status. They empirically support the key features of their model and find that the extent to which parents monitor their children decreases with socioeconomic status. Falk et al. (2021) also link socioeconomic status and parenting styles. They find that parents with low socioeconomic status more often resort to parenting with a higher degree of punishment and less often to warm and child-oriented parenting. In this sense, Weinberg (2001) argues that, because of the scarcity of means, low-income parents have limited access to create incentives for the child. Therefore, they more often resort to authoritarian methods, such as corporal punishment. Doepke & Zilibotti (2017) find similar evidence. They show that parental education is associated with a lower probability to be a neglecting or authoritarian parent. In contrast, the probability to be an authoritative parent increases.¹

¹Doepke & Zilibotti (2017) classify parents into Baumrind's (1991) four parenting styles using two questions asking children whether their parents are (i) supportive and (ii) strict/demanding.

Despite the notion of Becker et al. (1960) that a larger number of children tends to lower investment in each individual child, the link between the household composition and the choice of parenting style is less considered. For parents with more than one child, it may not be possible to follow a warm and child-oriented style due to constraints.

The third and last strand of literature we contribute to is about the link between parenting styles and children's cognitive and non-cognitive skills. In psychology, many studies have attempted to establish this relationship. Often, such studies focus on achievement in school, the child's personality or non-cognitive skills (see for example Aunola et al. 2000, Aunola & Nurmi 2005, Alegre 2011, Masud et al. 2015). Most commonly, they find that an authoritative style, or the features that an authoritative style exhibits, are associated with the most favorable outcomes. The economic literature provides similar evidence. Doepke & Zilibotti (2017) show that authoritative parenting is associated with better performance in school and higher educational attainment. Cobb-Clark et al. (2019) find a positive association between respectful parenting and the child's educational outcomes as well as on non-cognitive skills in youth (internal locus of control and less risky behavior). Higher parental monitoring is associated with less risky behavior. Fiorini & Keane (2014) and Falk et al. (2021) study the association between parenting styles and early childhood skills. Falk et al. (2021) analyze the link between the parenting style and the child's patience, risk aversion, behavior, altruism and IQ. They find positive effects of a warmer and more child-oriented parenting style on all these outcomes. Fiorini & Keane (2014) show that non-cognitive skills like behavioral problems, social skills, and emotional problems are especially sensitive to the parenting style. They find that a parenting style combining effective discipline and parental warmth, i.e. an authoritative style in the sense of Baumrind, leads to the most favorable non-cognitive outcomes. In contrast to Falk et al. (2021), they find that cognitive skills are less sensitive to the parenting style.

We contribute to the literature in many ways. First, we apply a novel method which can handle a large set of measures on parent-child interactions. Therefore, we are able to describe parenting styles in more detail than previous studies. This allows us to separate styles that differ only in terms of a few, but important, dimensions. Second, the theoretical framework of LDA-S provides an economically interpretable link between parent-child interactions and parents' socioeconomic environment. Third, in contrast to continuous latent factors, latent classes more easily refer to theoretical models such as Baumrind (1971, 1991). In this way, the data driven approach can be embedded into theoretical frameworks. Fourth, we fill the gap on the link between parenting styles and household composition. Fifth, rich data on children's (non-)cognitive skills allow us to explore the association between parenting styles and children's skills. Finally, we are able to analyze the role of parenting styles in the emergence of skill gaps between children from different socio-economic environments in early childhood.

Applying LDA-S results in four parenting styles. Two styles closely resemble Baumrind's (1991) authoritative and authoritarian style. The two other styles can be interpreted as variations. One style is very similar to an *authoritative* style, which we call *democratic-loving*. The democratic-loving style differs from the authoritative style as such parents do not enforce their will, leave more autonomy to their child and communicate with the child more positively. The last style is like an *authoritarian* style, but the parents are much more inconsistent in their parenting. We call this style *authoritarian-inconsistent*. Our results show that parenting styles are strongly associated with household income, education and whether the child is an only child. Although our model does not directly allow for the identification of potential channels, the results suggest that constraints in both time and (non-)cognitive skills of the parents play an important role in choosing a parenting style. We find that children's (non-)cognitive skills are strongly associated with the parenting style. In line with Fiorini & Keane (2014), this link is more pronounced for non-cognitive than for cognitive skills. An *authoritative* and a *democratic-loving* style are associated with the highest skills, whereas children who are raised with an *authoritarian-inconsistent* style have the lowest skills. Our results show how differences in parenting styles contribute to the skill gap between children from different socioeconomic environments. We find that in particular styles associated with low household income are linked with lower skills. Further, parents with more than one child are more likely to choose a style that is related to lower skills. In contrast, having an only child is associated with a style that is associated with higher skills. Interestingly, parents' education is not systematically connected to parenting styles which are related to more favorable outcomes.

The remaining paper is structured as follows. Section 2 briefly describes the data. In section 3, we describe the method applied in the empirical analysis. Section 4 describes the parenting styles which are identified by our model. In section 5, we show how these parenting styles are linked with parental socioeconomic environment. Section 6 presents and discusses how the parenting styles are linked to (non-)cognitive skills. Section 7 concludes.

2 Data

This paper uses the first Starting Cohort (NEPS-SC1) of the German National Educational Panel Study (NEPS 2021*b*). The panel study follows children born between February and July 2012 since they were six months old. One parent of every child is interviewed as part of the study. The data is perfectly suited to answer our research questions. It contains extensive information on each child, the child's development, the household in which the child lives in as well as on parents and how they rear their child. Our analysis mainly relies on questions about the parent-child interaction and measures on (non-)cognitive skills.

2.1 Parent-Child Interaction

To identify parenting styles, we rely on J = 23 questions about the parent-child interaction when the child was 5 and 6 years old. Broadly, the parent-child interaction can be classified into six categories: (1) How parents monitor their child, (2) how parents enforce their will, (3) how inconsistent parents are in their parenting, (4) how emotionally warm parents are with their child, (5) how parents communicate with their child and (6) how much autonomy parents leave to their child. Table 1 summarizes the questions and shows the response behavior of parents from 1530 children. For most of the questions, there is a rather large amount of response heterogeneity. The distribution is mostly concentrated around one option, i.e. multiple mass points at extremes of the response distribution do not exist. Our goal is to link parenting styles with household income, parental education and household composition. To this end, we compute the monthly household equivalence income (Hagenaars et al.

1994) and split it into three categories using the 33%- and 66%-quantile of its distribution. We use an indicator showing whether at least one parent has a university degree to measure parental education and an indicator that shows whether the child is an only child to measure the household composition. In the right panel of table 1 we depict p-values of a Chi-Square Test that tests the null of independence between parent-child interaction x_{ij} and the parental characteristics. Results show that there are substantial differences in parenting between parents with different characteristics. In section 5, we analyze whether there are any patterns behind these differences.

	never	seldom	some- times	often	very often	χ^{inc}	$\chi^{\rm educ}$	χ^{sib}
Monitoring of Parents (M)								
If your child has new friends, you talk to him/her about them.	0.00	0.03	0.17	0.56	0.24	0.01	0.01	0.26
If your child went out, you ask him/her what he/she did and experienced.	0.00	0.00	0.01	0.28	0.71	0.05	0.79	0.75
If your child's out, you know exactly where he/she is.	0.00	0.00	0.01	0.11	0.88	0.18	0.46	0.92
If your child has new friends, you will meet them soon.	0.00	0.01	0.09	0.37	0.53	0.04	0.17	0.25
Enforcement of Will (E)								
When your child starts to negotiate with you, you exercise your authority.	0.01	0.11	0.52	0.33	0.03	0.59	0.00	0.11
You set clear limits for your child so that it does not exploit your goodwill.	0.00	0.04	0.24	0.59	0.12	0.01	0.00	0.68
If you want your child to do something, you give a clear command and don't	0.01	0.11	0.35	0.47	0.06	0.39	0.00	0.48
tolerate any great detours.								
If your child wants you to make an exception, you insist on your rules so that	0.05	0.23	0.50	0.20	0.02	0.06	0.01	0.67
it is clear who is in charge in the family.								
Inconsistency of parenting (I)								
You soften a punishment or terminate it prematurely.	0.04	0.33	0.49	0.12	0.02	0.05	0.34	0.11
On some days you are stricter with your child than on the others.	0.01	0.18	0.69	0.11	0.01	0.20	0.04	0.51
You threaten to punish your child, but you don't punish him/her.	0.18	0.48	0.28	0.06	0.00	0.00	0.02	0.02
It's hard for you to be resolute in your parenting.	0.09	0.47	0.37	0.06	0.01	0.03	0.00	0.05
Emotional Warmth (W)								
You show your child with words and gestures that you love him/her.	0.00	0.00	0.02	0.27	0.71	0.05	0.01	0.30
You comfort your child, when it is sad.	0.00	0.00	0.02	0.30	0.68	0.77	0.46	0.74
You praise your child.	0.00	0.00	0.03	0.56	0.41	0.39	0.00	0.14
Communication of parents (C)								
You criticize you child.	0.04	0.30	0.56	0.09	0.00	0.00	0.00	0.28
You shout at your child, if he/she has done something wrong.	0.15	0.55	0.27	0.03	0.00	0.17	0.19	0.04
You insult your child when you are angry at him/her.	0.46	0.34	0.18	0.02	0.00	0.10	0.06	0.04
	does not	does rather	does r	ather	does completely	xinc	χ^{educ}	χ^{sib}
Autonomy of child (A)	apply at all	not apply	app	bly	apply	χ^{me}	χ^{cuuc}	χ^{310}
I think it's good if my child says what it thinks.	0.00	0.01	0.36		0.64	0.49	0.03	0.93
If my child wants something and doesn't get it, I'll explain why.	0.00	0.01	0.28		0.72	0.24	0.17	0.34
I often ask my child for opinion.	0.00	0.10	0.61		0.28	0.69	0.00	0.01
I let my child make its own plans for the things it wants to do.	0.01	0.17	0.66		0.15	0.88	0.00	0.34
If I want my child to do something, I'll explain why.	0.00	0.06	0.55		0.39	0.67	0.10	0.24

Table 1 - Measures on Parent-child interaction

The table summarizes the measures on parent-child interaction. The panel in the middle depicts the survey responses. The right panel shows the p-values of a Chi-Square Test testing the null of independence between each parent-child interaction and monthly equivalence household income in three categories (χ^{inc}), whether at least parent has a university degree (χ^{educ}) or whether the child is an only child (χ^{sib}).

2.2 Non-cognitive and Cognitive Skills

The final part of this paper analyzes how parenting styles affect (non-)cognitive skills of the child. NEPS-SC1 collects an extensive set of different skill measures. In our main analysis, we focus on outcomes that were surveyed when the child was 4 and 7 years old. Cognitive skills are assessed via standardized tests (Berendes et al. 2013, NEPS 2020, 2021*a*). The measurement of basic cognitive skills is based on tests that are as education-independent and domain-unspecific as possible. To measure mental performance, we rely on the child's ability to reason. Linguistic skills are undisputedly very import determinants for explaining social disparities in school careers. These are captured via listening comprehension. To test the mathematical literacy, the child is required to recognize and flexibly apply mathematics in realistic, mainly extra-mathematical situations.

To analyze the effect of parenting styles on non-cognitive skills, we use the Goodman's (1997) *Strengths and Difficulties Questionnaire* (SDQ) to measure social behavior (see Wohlkinger et al. 2019). We also observe the patience of the child. It is measured by a classical experiment on the delay of gratification, where the child could choose between one gift now or two gifts tomorrow (NEPS 2021*a*).

In addition, we conduct supplementary analyses for outcomes that are only surveyed once. We analyze the effect of parenting styles on the child's personality traits measured by Big Five and how children cope with their every-day school life. This includes the child's autonomy, enjoyment of learning, willingness to make an effort, and social integration into the class.

3 Latent Dirichlet Analysis for Survey Data

Motivated by the differences shown in table 1, our goal is to explain the heterogeneity in parent-child interaction given the parental education, household income and whether the child is an only child. We apply an adapted version of Latent Dirichlet Analysis (see Blei et al. 2003) for Survey Data (LDA-S) proposed by Munro & Ng (2022). LDA-S connects unobserved heterogeneity with observed characteristics and survey responses, explicitly acknowledges that survey responses are categorical and provides an economic interpretation of the unobserved heterogeneity. Throughout the paper, italic symbols denote scalars and bold symbols denote vectors that collect the respective scalars along their indices.

Assume we observe N parents indexed by i. Each parent belongs to one of $d_i \in \mathbb{G} = \{1, ..., G\}$ observable groups. In our case, individuals are grouped by all possible combinations of three categories of household equivalence income, an indicator that shows whether one parent has a university degree and whether the child has siblings living in the same household, i.e. G = 12. We observe J dimensions of the interaction between the target child of the survey and the parents x_{ij} . Each dimension j has L^j possible responses, where parents choose a single response v from $x_{ij} \in \mathbb{L}^j = \{1, ..., L^j\}$. We model the heterogeneous parent-child interaction as coming from K possible strategies to raise a child $z_i \in \mathbb{K} = \{1, ..., K\}$ (i.e. parenting styles). Parents choose z_i such that their utility is maximized. The model incorporates a group-affinity, which allows parents with similar income, education and number of children to choose the same parenting style. An individual effect allows parents to deviate from their group affinity, though.

$$z_{i} = \underset{k \in 1, ..., K}{\arg \max} U(k) = \underset{k \in 1, ..., K}{\arg \max} \sum_{j}^{K} \mathbb{1} \left(k = j \right) \left(u_{d_{i}, j} + e_{ij} \right),$$
(1)

where $u_{d_i,j}$ denotes the group affinity of $d_i = g$ for style j = k and e_{ij} is an individual effect that captures everything else. The observed heterogeneity of an individual's group membership d_i and unobserved heterogeneity of an individual's chosen parenting style is linked by a random variable that gives the probability to choose parenting style $z_i = k$ given group membership $d_i = g$

$$\pi_{gk} = \mathbb{P}\left(z_i = k | d_i = g\right) = \mathbb{P}\left(u_{gk} + e_{ik} = \max_{j \in \mathbb{K}}\left(u_{gj} + e_{ij}\right)\right).$$
(2)

The chosen parenting style influences the observed parent-child interaction x_{ij} . Parents optimally interact with their child by maximizing their individual score function for each survey question.

$$x_{ij} = \arg\max_{v \in \{1,...,L^j\}} \sum_{u=1}^{L_j} \mathbb{1}\left(v = u\right) \left(q_{z_i,u}^j + s_{iu}^j\right),\tag{3}$$

where $q_{z_i,u}^j$ denotes a parenting style-specific effect and s_{iu}^j an individual-specific effect, which allows parents to deviate from the usual parent-child interaction with parenting style $z_i = k$. The expected parent-child interaction is described by a random variable $\beta_{k,v}^j$ that captures the probability that an individual with parenting style $z_i = k$ chooses v as the response to survey question j.

$$\beta_{kv}^{j} = \mathbb{P}\left(x_{ij} = v | z_i = k\right) = \mathbb{P}\left(q_{z_i,v}^{j} + s_{iv}^{j} = \max_{u \in \mathbb{L}^{j}}\left(q_{z_i,u}^{j} + s_{iu}^{j}\right)\right)$$
(4)

Since we neither observe $\boldsymbol{u}_{g,:}, \boldsymbol{e}_{i,:}, \boldsymbol{q}_{k,:}^{j}$ and $\boldsymbol{s}_{i,:}^{j}$ nor their distribution, $\boldsymbol{\pi}_{g,:}$ and $\boldsymbol{\beta}_{k,:}^{j}$ are treated as random. Each $\boldsymbol{\pi}_{g,:}$ and each $\boldsymbol{\beta}_{k,:}^{j}$ is multinomial distributed. Therefore, Munro & Ng (2022) specify priors that follow a Dirichlet distribution with hyperparameters $\boldsymbol{\alpha}_{g,:} \in \mathbb{R}^{K}$ and $\boldsymbol{\eta}_{k,:}^{j} \in \mathbb{R}^{L^{j}}$ respectively. In summary, LDA-S is identified by a hierarchical model

$$x_{ij}|m{eta}, z_i \sim \mathsf{Multinomial}\left(m{eta}_{z_i,:}^j
ight)$$
 (5a)

$$z_i | \boldsymbol{\pi}_{d_i,:} \sim \mathsf{Multinomial}\left(\boldsymbol{\pi}_{d_i,:}\right)$$
 (5b)

$$\pi_{d_{i},:} \sim \mathsf{Dirichlet}\left(\boldsymbol{\alpha}_{d_{i},:}
ight)$$
 (5c)

$$eta^{j}_{z_{i},:}\sim \mathsf{Dirichlet}\left(oldsymbol{\eta}^{j}_{z_{i},:}
ight).$$
 (5d)

Using this, we can write down the joint distribution and estimate the model using MCMC methods (see Munro & Ng 2022). The Gibbs Sampler iteratively samples each variable from its conditional distribution conditional on all other variables. Each iteration comprises three steps. First, z_i conditional on $\mathbf{x}_{i,:}$, β , and $\pi_{d_i,:}$ is sampled form a multinomial distribution. Second, β conditional on η , \mathbf{x} , and \mathbf{z} is sampled form a Dirichlet distribution. Third, $\pi_{g,:}$ conditional on α , \mathbf{x} , and \mathbf{z} is sampled form a Dirichlet distribution. Throughout the process, new values of the variables are used as soon as they are obtained. Draws of z_i depend on the values of β and $\pi_{d_i,:}$ from the previous iteration, whereas draws of β and $\pi_{g,:}$ depend on z_i from the current one. In our analysis, we conduct 20000 iterations. The results shown in sections 4, 5 and 6 are based on the the sample averages over the whole process, as it is usually done. To account for the bias caused by starting the system with randomly chosen initial values (initial transient), we burn the first 10000 iterations.

To estimate the model, hyperparameters of the Dirichlet distributions, $\alpha_{g,:}$ and $\eta_{k,:}^j$, and the number of parenting styles, K, have to be specified. Hyperparameters of the Dirichlet prior specify the researcher's beliefs about the importance of the group-specific terms ($u_{g,:}$ and $q_{k,:}^j$) relative to the individual-specific ones ($e_{i,:}$ and $s_{i,:}^j$). For example, we would specify $\alpha_{g,k} < 1$ if we believe that members of the same observable group g are likely to choose the same parenting style k. Similarly, $\eta_{kv}^j < 1$ reflects the belief that all individuals who choose the same parenting style, are likely to respond the same way. The opposite is true for $\alpha_{g,k} > 1$ and $\eta_{kv}^j > 1$. We impose uninformative priors to the relationship between observed group affinity and parenting style or parenting style and response behavior, i.e. $\alpha_{g,k} = 1 \forall g, k$ and $\eta_{kv}^j = 1 \forall j, k, v$. Regarding the number of parenting styles, we

follow Munro & Ng (2022) and choose the optimal K according to the minimum of an approximated Bayesian information criterion (BIC). In our case K = 4.

In summary, LDA-S imposes structure on observable group indicators and parents' responses in the questionnaires by assuming that parents optimally choose parenting styles given their group membership and optimally select responses given their chosen parenting styles. The optimal choices are affected by individual terms and group commonalities in the first or parenting style commonalities in the second case. The individual effects allow parents to deviate from the choices usually made by other parents with the same group or parenting style.

4 Identification of Parenting Styles

In this section, we present the parenting styles defined by LDA-S. Our results show that Parenting Style 1 is chosen slightly more often (31%) than Parenting Style 2 (28%). Parenting Style 3 and 4 are chosen less often (20%). We want to give each parenting style a meaningful interpretation. Figure 1 and 2 depict the probability for an individual with parenting style $z_i = k$ to choose v as response to survey question j, i.e. $\beta_{k,:}^j$.

The far left area of figure 1 shows how the typical **monitoring** behavior of parents, given their parenting style, looks like. Parents who choose Style 1 state (1) with a probability of 0.40 that they talk about the child's new friends very often, (2) with a probability of 0.93 that they ask about the child's experiences very often, (3) with a probability of 0.95 that they know where the child is very often and (4) with a probability of 0.69 that they meet their child's new friends very often. Typical parents with Style 4 behave similarly. In contrast, the respective probabilities of parents who choose Style 2 or Style 3 are much smaller. The left area of figure 1 shows that the styles also differ in the way how parents enforce their will. Parents with Style 1, Style 2, or Style 3 often enforce their will with a rather high probability. In contrast, the respective probabilities are much smaller for parents with Style 4. The parent-child interaction along emotional warmth shows that parents who choose Style 1 or Style 4 are likely to be very warm in their parenting. In contrast, parents with Style 2 or Style 3 are emotionally warm with a much smaller probability. With regard to inconsistent parenting, parents who choose Style 3 stand out. They state (1) with a probability of 0.34 that they often soften a punishment, (2) with a probability of 0.27that they often are stricter on some days, (3) with a probability of 0.22 that they often inconsistently threaten their child and (4) with a probability of 0.21 that it is *often* hard for them to be resolute in their parenting. The respective probabilities for Style 1, Style 2, or Style 4 are very close to zero. The parent-child interaction along the dimension of communication shows that typical parents who choose Style 4 communicate with the child in a negative way with a very low probability. They state (1) with a probability of 0.45 that they seldom criticize the child, (2) with a probability of 0.33 that they *never* shout at the child and (3) with a probability of 0.65 that they *never* insult the child. In contrast, parents with Style 3 state (1) with a probability of 0.22 that they *often* criticize the child, (2) with a probability of 0.53 that they sometimes shout at the child and (3) with a probability of 0.37 that they sometimes insult the child. The respective probabilities of Style 1 and Style 2 are somewhere between those of Style 3 and Style 4.



Figure 1 – Probability to respond given choice of parenting style I

The figure depicts $\beta_{k,i}^{j}$, i.e. the probability for an individual with parenting style $z_{i} = k$ to choose v as response to survey question j.

In figure 2, we show how much **autonomy** parents leave to their child. The behavior of parents with Style 4 stands out. They typically leave their child much autonomy. Children who are raised with Style 2 and 3 are likely to be less autonomous. The probabilities for parents with Style 1 lie in between. Their child is likely to be more autonomous than those of Style 2 or Style 3 but less than those of Style 4.

Figure 1 and 2 show major differences between the four parenting styles. To summarize these differences, we compute the Rao distance between $\beta_{k,:}^{j}$ and $\beta_{m,:}^{j}$ for all $k \neq m$ (Munro & Ng 2022). Table 2 depicts the five dimensions of parent-child interaction where the parenting styles differ most from each other for each parenting style.

The results show that talking about new friends and asking what the child experienced are the two biggest differences between Style 1 and Style 2. In addition, four out of the five biggest differences between Style 1



Figure 2 - Probability to respond given choice of parenting style II

The figure depicts $\beta_{k,i}^{j}$ i.e. the probability for an individual with parenting style $z_{i} = k$ to choose v as response to survey question j.

and Style 4 can be followed back to the way parents enforce their will. We conclude that typical parents who choose Style 1 monitor their child, are consistent, powerfully enforce their will yet leave the child autonomy and are emotionally warm. This style closely mirrors Baumrind's (1991) *authoritative* style.

Besides the major difference between Style 1 and Style 4 in that parents with Style 4 do not powerfully enforce their will, one of the key difference between Style 4 and Style 1 in table 2 is shouting at the child. This is also one of the main differences between Style 4 and Style 3. Thinking that it's good if the child says what she thinks, explaining why the child doesn't get something and asking the child for her opinion are among the five biggest differences between Style 4 and Style 2. The latter also belongs to the biggest differences between Style 4 and Style 2. The latter also belongs to the biggest differences between Style 4 and Style 4 is similar to Style 1 in many aspects. They differ, as parents with Style 4 typically do not enforce their will but leave their child more autonomy, and do not communicate negatively. This style is not only closely related to Baumrind's (1991) *authoritative* style, but also to Baumrind's (1991) *permissive* style.

	Parenting Style 2	Parenting Style 3	Parenting Style 4
	(M) Talk about new friends	(I) Threaten child	(E) Set clear limits
	(M) Ask what child experienced	(I) Hard to be resolute	(E) Exercise authority
Parenting Style 1	(W) Praise child	(W) Praise child	(E) Give a clear commands
	(E) Set clear limits	(I) Soften a punishment	(E) Insist on rules
	(W) Gestures	(E) Set clear limits	(C) Shout at child
		(I) Threaten child	(A) Ask child's opinion
		(I) Hard to be resolute	(A) Good if child says what it thinks
Parenting Style 2		(I) Soften a punishment	(M) Talk about new friends
		(C) Shout at child	(A) Explain why child doesn't get sth.
		(I) On some days stricter	(W) Praise child
			(I) Threaten child
			(C) Shout at child
Parenting Style 3			(I) Hard to be resolute
			(A) Good if child says what it thinks
			(E) Exercise authority

Table 2 - Largest Differences Between Parenting Styles

The table summarizes the five biggest differences between each parenting style. Differences are computed using the Rao distance.

However, *permissive* parents do not extensively monitor their child. As positive and participative communication distinguish this style, we define Style 4 as *democratic-loving*.

Style 2 strongly differs from Style 4 in talking about new friends. Further, table 2 underlines that parents with Style 2 are emotionally much colder than parents with Style 1 or Style 4. Showing love with words or gestures belongs to the biggest differences between Style 1 and Style 2. Praising the child is one of the biggest differences between Style 4 and Style 2. We conclude that parents who choose Style 2 powerfully enforce their will, are not as emotionally warm as *authoritative* or *democratic-loving* parents and typically do not take the child's will into account as much as *authoritative* or *democratic-loving* parents. In line with Baumrind (1991), we call such parents *authoritarian*.

Table 2 also shows that differences between Style 3 and the other styles are mainly due to inconsistent behavior. Other than that, Style 3 closely mirrors an *authoritarian* style. Therefore, we refer to this style as *authoritarian-inconsistent*.

5 Parenting Styles and Socio-Economic Environment

By modeling group affinities for parenting styles, our model directly incorporates differences in parenting styles along parents' education, household equivalence income and whether the child is an only child. In this section, we interpret $\pi_{g,:}$, i.e. the probability to choose style k given membership of observable group g. Table 3 shows the average probabilities of $\pi_{g,:}$ for each parental characteristic separately.

The results show that parental education is an important determinant in choosing a parenting style. On average, parents with a university degree are more likely to raise their child with an *authoritarian* or *democratic-loving* style than parents without a university degree. In contrast, they are less likely to choose an *authoritative* or

	Authoritative	Authoritarian	Authoritarian- inconsistent	Democratic- loving	Number of Observations
Education					
No University	0.34	0.26	0.22	0.18	643
University	0.27	0.31	0.19	0.23	887
Household equ	ivalence income				
Low	0.26	0.28	0.25	0.21	431
Middle	0.29	0.30	0.21	0.20	504
High	0.34	0.28	0.17	0.21	595
Siblings					·
Only child	0.32	0.23	0.21	0.25	323
Siblings	0.30	0.30	0.20	0.20	1207
				·	

Table 3 – Average probabilities

The table depicts the average probabilities to choose each style for each parental characteristic separately. Probabilities are computed by averaging $\pi_{g,:}$ along observable groups weighted by the number of observations in each observable group.

authoritarian-inconsistent style. The average probability to choose an *authoritative* style becomes higher with rising household income. In contrast, it is less likely to raise the child with an *authoritarian-inconsistent* style for parents with higher household income. The average probability to choose an *authoritarian* or *democratic-loving* style is not associated with household income. The household composition is strongly associated with the average probability to choose an *authoritarian* and *democratic-loving* style. Whereas the probability to raise the child with an *authoritarian* style is smaller for parents with an only child than for parents with more than one child. The probability to choose a *democratic-loving* style is higher for parents with an only child.

Figure 3 shows the link between parenting style and parents' socioeconomic status as well es as household composition in more detail. It depicts the probability to choose a parenting style given the membership of observable group g. An *authoritative* style is most likely chosen by parents with more than one child, without a university degree and with high household income (0.46). The probability becomes smaller for lower income, however stays on a high level (0.29 and 0.34). For parents with lower education and an only child the probabilities to choose an *authoritative* style are also high (upper left panel). Those with low income have the lowest probability, although differences between income levels are less pronounced. The lower panels reveal that parents with low income and university degree are less likely to raise their child *authoritatively*. The probability becomes larger for higher income.

Parents with a university degree and more than one child are very likely to choose an *authoritarian* style (lower right panel). The probability is the highest if the income of the household is low (0.45). For parents with a middle and a high income the probability is much smaller, but stays on a high level (0.3). In comparison, parents with a university degree and an only child are less likely to choose an *authoritarian* style, especially if the income of the household is small (0.06) or in the middle (0.2). Opposed to highly educated parents with more than one child, the probability becomes higher with increasing income. For parents with lower education and an only child, the probability also systematically increases with higher income. The level of the probability is consistently higher compared to parents with a university degree and an only child, but smaller compared to parents without university education, more than one child and a low or middle household income. Thus, figure 3 shows that the



Figure 3 – Probability to choose parenting style k given membership of observable group The figure shows $\pi_{g,:}$, i.e. the probability to choose style k given membership of observable group g.

number of children is strongly associated with the decision to choose an *authoritarian* style. Parents with an only child are less likely to choose such a style. The role of education and income in choosing an *authoritarian* style clearly differs with the number of children.

The probability to choose an *authoritarian-inconsistent* style does not vary by observable groups as much as the probability to choose one of the other styles. Mostly, the probability lies between 0.2 and 0.28. However, we find a clear pattern, where parents with lower income are systematically more inclined to choose this style.

Among all parents, those with one child and a university degree have the highest probability to choose a *democratic-loving* style (lower left panel). The probability is especially high for a low household income (0.5) and decreases for parents in households with middle or high income (0.4 and 0.26 respectively). Highly educated parents with more than one child or parents with low education are less inclined to raise their child with a *democratic-loving* style.

In summary, average probabilities depicted in table 3 show a strong association between parenting styles and parental characteristics. However, figure 3 indicates, that the link between parenting styles, parental socioeconomic

status and household composition is complex. The probabilities strongly differ along observable groups, i.e. the combination of all three variables. The results show that there is no style which is clearly preferred by one group. This indicates that other characteristics not captured by the observable groups are very important determinants.

The results point to the importance of parents' time resources and (non-)cognitive skills when choosing a parenting style. Parents who choose a democratic-loving style strongly focus on the needs of the child and do not impose their will by directly restricting the child's actions. Such parents rather enforce their will by persuading the child. In contrast, an authoritative or authoritarian style directly restrict the child's actions. At the same time, holding a university degree is a good predictor for choosing a democratic-loving style. An authoritative style is generally more likely chosen by parents without a university degree. For parents with an only child, an authoritarian style is more likely chosen by parents with lower education. Therefore, we conclude that a democratic-loving style demands high (non-)cognitive skills of the parents. In contrast, an authoritative or an authoritarian style are not as (non-)cognitively demanding as a democratic-loving style. Therefore, (non-)cognitive skills may play an important role in the choice of a parenting style. Further, a democratic-loving style requires to discuss issues in case of disagreement between child and parent or to let the child make own plans. In contrast, an authoritarian style leaves less autonomy to the child. The number of children indicates whether parents are able to give their full attention to only one child or whether they have to allocate their time to multiple children. The results of the model show that parents with more than one child are less likely to choose a democratic-loving style, but more likely choose an authoritarian style. Therefore, we conclude that a democratic-loving style requires more time resources compared to an authoritarian style. This underlines the importance of time resources in choosing a parenting style.

6 Parenting Styles and Children's Skill Development

6.1 Estimation Strategy

To analyze how parenting styles are related to the skill development of children, we estimate a simple static model, where we look at one specific period in childhood. Our data contains M skills S_i^{7m} when the child was 7 years old indexed by m. Our model uses predictions $z_{i1}, ..., z_{i4}$ for the parenting style of parents i derived from the estimation in section 4 and 5 (z_{ik} equals one for the most likely parenting style for parents i and zero otherwise). We assume that S_i^{7m} is linearly affected by the parenting style z_{ik} , the initial endowment of that skill S_i^{4m} at age 4, and a vector of child's and parent's individual characteristics captured by \mathbf{X}_i^m .

$$S_i^{7m} = \beta_{0g}^m + \sum_{k \in 2,3,4} \gamma_{1k}^m z_{ik} + \delta_1^m S_i^{4m} + \boldsymbol{\delta}_2^{m'} \boldsymbol{X}_i^m + \eta_i^m,$$
(6)

where the base category is z_{i1} , i.e. whether *i* chooses an *authoritative* style or not, and β_{0g}^m is an intercept that varies with the observed group membership. The coefficient γ_{1k}^m can be interpreted as the effect of the parenting style on the change in skills between age 4 and 7. This effect is biased if, given X_i^m , unobserved factors affect both the choice of the parenting style and the change in skills of the child. For example, the speed at which children

learn may be related to factors that are also associated with the parents' choice of parenting style (genetics, neighborhood, etc.). Unfortunately, we cannot use an instrumental variable approach to solve this issue, since we could not find any exogenous source of variation that affects the choice of the parenting style. Therefore, we use an extensive set of control variables to mitigate the bias induced by potential confounding factors. First, other skills at age 4 may affect both, the change in skills of the child and the choice of parenting style. Hence, X_i^m contains the initial endowment of all other considered skills and additional measures on (non-)cognitive skills at age 4 (e.g. measures on the child's temperament). Second, the choice of the parenting style may depend on parental skills and preferences, which may also have a strong effect on the change in the child's skills. To account for this, we control for the respondents personality traits, patience and risk aversion. Third, peers may be important confounding factors. Therefore, we control for the share of parents' friends who hold a university degree, the share of parents' friends with migration background and the share of the child's friends with migration background. Fourth, the quality time parents spend with their child may be correlated with both, the change in skills of the child and the choice of the parenting style. To address this source of bias, we control how much quality time parents spent with their child when the child was 6 years old, i.e. in our considered period of childhood between age 4 and 7. The indicator is constructed by taking the average over how often parents (1) read a story to their child, (2) show single letters or the alphabet to the child, (3) practice numbers with the child, (4) teach short poems, rhymes or songs to the child, (5) paint, draw or craft with the child, (6) go to the library with the child, and (7) tell a story to the child. Finally, we control for demographic characteristics of the child and the parents. All controls are summarized in table A1 in the appendix.

Using equation (6), we estimate how parenting styles are related to cognitive skills (mathematical literacy, listening comprehension, reasoning) and non-cognitive skills (problem behavior, prosocial behavior and patience). We also report supplementary results on how parenting styles are associated with the child's personality traits measured by the Big Five. However, the child's initial endowments of personality traits at age 4 are not available in the data due to the children's young age. Since parents pass on their skills and preferences to their child through genetic, social or other channels, we use the personality traits of the interviewed parent to measure the child's initial endowment (comparable to Falk et al. 2021). Further, we analyze how the children cope with their school day. No initial endowments can be observed since children were recently enrolled.

6.2 Results

We estimate equation (6) using ordinary least squares. All regression models contain the same control variables. The outcomes are normalized to have mean 0 and standard deviation 1. Table 4 shows the main results of our analysis.

The upper panel shows the effect of the parenting style on cognitive skills. The ability to reason is not significantly associated with the parenting style. A *democratic-loving* style is associated with a significant higher mathematical literacy compared to an *authoritarian* or *authoritarian-inconsistent* style. Further, the listening comprehension of children raised with a *democratic-loving* style is significantly higher than the listening comprehension of children

P	uthoritative vs		Authoritari	ian vs	Authoritarian- inconsistent vs	
A	Authoritarian-	Democratic-	Authoritarian-	Democratic-	Democratic-	N
Authoritarian	inconsistent	loving	inconsistent	loving	loving	
-0.01	0.07	0.00	0.08	0.02	-0.07	1373
0.05	0.09	-0.08	0.03	-0.13**	-0.17**	1364
-0.02	0.12	-0.14*	0.14*	-0.12	-0.26***	1228
			-			
0.10	0.30***	-0.08	0.19**	-0.18**	-0.38***	1182
-0.13*	-0.17**	-0.01	-0.04	0.12	0.16*	1184
0.13*	0.07	0.23***	-0.07	0.10	0.16*	1227
	0.05 -0.02 0.10 -0.13*	Authoritarian inconsistent -0.01 0.07 0.05 0.09 -0.02 0.12 -0.01 0.30*** -0.13* -0.17**	Authoritarian Inconsistent Ioving -0.01 0.07 0.00 0.05 0.09 -0.08 -0.02 0.12 -0.14* 0.10 0.30*** -0.08 -0.13* -0.17** -0.01	Authoritarian Inconsistent Ioving Inconsistent -0.01 0.07 0.00 0.08 0.05 0.09 -0.08 0.03 -0.02 0.12 -0.14* 0.14* 0.10 0.30*** -0.08 0.19** -0.13* -0.17** -0.01 -0.04	Authoritarian inconsistent loving inconsistent loving -0.01 0.07 0.00 0.08 0.02 0.05 0.09 -0.08 0.03 -0.13** -0.02 0.12 -0.14* 0.12 -0.12* 0.10 0.30*** -0.08 0.19** -0.18** -0.13* -0.17** -0.01 -0.04 0.12	Authoritarian- inconsistent Democratic- loving Authoritarian- inconsistent Democratic- loving Democratic- loving -0.01 0.07 0.00 0.08 0.02 -0.07 0.05 0.09 -0.08 0.03 -0.13** -0.17** -0.02 0.12 -0.14* 0.14* -0.12 -0.26*** 0.10 0.30*** -0.08 0.19** -0.18** -0.38*** -0.13* -0.17** -0.01 -0.04 0.12 0.16*

Table 4 - Parenting Styles and Skills - Main Results

The table depicts the effect of each parenting style on child's (non-)cognitive skills. All controls are summarized in table A1. Controls include the initial endowment of the skill observed at age 4. Outcomes are normalized to have mean 0 and standard deviation 1. Significance of the coefficients at conventional significance levels 1%, 5%, 10% are indicated by stars ***, **, *, respectively. The last column N shows the number of observations.

with *authoritarian-inconsistent* or *authoritative* parents. Children who are raised with an *authoritarian* style have a higher listening comprehension compared to children raised with an *authoritarian-inconsistent* style.

In the lower panel, we report how parenting styles are related to non-cognitive skills. Children raised by *authoritarian-inconsistent* parents are less prosocial than children with *authoritative, democratic-loving* or *authoritarian* parents. Further, they exhibit problem behavior more frequently than children with parents who choose an *authoritative* or a *democratic-loving* style. An *authoritarian* style is associated with less prosocial behavior than a *democratic-loving* style and with a more frequent problem behavior than an *authoritative* style. Regarding social behavior (i.e. prosocial and problem behavior) *authoritative* and *democratic-loving* styles do not differ. However, we find that children who are raised by *authoritative* parents are more patient than those raised by *democratic-loving* parents. They are also significantly more patient than children of *authoritarian* parents. Children raised with *authoritarian-inconsistent* style are significantly more patient than children with *democratic-loving* parents.

Table 5 summarizes supplementary results on how parenting styles are related to the child's non-cognitive skills.

The results in the upper panel show how the child's personality traits are associated with the parenting style. Children who are raised with an *authoritative* or a *democratic-loving* style are more conscientious, more agreeable, more open and less neurotic than children who are raised with an *authoritarian-inconsistent* style. Children with *authoritarian* parents are less open than children of *democratic-loving* or *authoritative* parents, but more agreeable and more conscientious than children who are raised with an *authoritarian-inconsistent* style. An *authoritative* style is associated with a higher extraversion compared to all other styles.

In the remaining panels, we analyze how parenting styles are related to how children cope with their everyday school life. Children with *authoritarian-inconsistent* parents are less autonomous, have less pleasure in learning, show less willingness to make an effort and are worse integrated in the class than those with *authoritative* or *democratic-loving* parents. Further, they are also less autonomous, have less fun in school and show less willingness to make an effort than children who are raised with an *authoritarian* style. Compared to an *authoritarian* style, children with *authoritative* parents can cope with many tasks more easily, have fun studying more often, try hard if a task is difficult more often and have many friends in class more often. Children with *democratic-loving* parents treat

		Authoritative vs		Authoritar	ian vs	Authoritarian- inconsistent vs	
	Authoritarian	Authoritarian- inconsistent	Democratic- loving	Authoritarian- inconsistent	Democratic- loving	Democratic- loving	N
Personality							
Extraversion	0.21***	0.18**	0.20**	-0.02	-0.01	0.01	1203
Conscientiousness	0.05	0.28***	-0.02	0.23***	-0.07	-0.30***	1202
Agreeableness	0.05	0.28***	-0.07	0.23***	-0.12	-0.35***	1191
Openness	0.19***	0.18**	0.02	-0.01	-0.17**	-0.16*	1202
Neuroticism	-0.09	-0.18**	0.02	-0.09	0.12	0.20**	1202
Coping with school day: Child's au	itonomy						
Doing homework independently	0.03	0.25***	0.00	0.22**	-0.03	-0.24**	1084
Needs support with homework	0.03	-0.28***	-0.01	-0.32***	-0.04	0.28***	1123
Can cope with many tasks easily	0.13*	0.20**	0.04	0.07	-0.09	-0.16*	1261
Coping with school day: Enjoymen	t of learning						
Likes to go to school	0.02	0.05	-0.08	0.03	-0.10	-0.13	1275
Having fun at school	0.06	0.20**	-0.04	0.14*	-0.10	-0.24**	1274
Having fun studiying	0.12*	0.19**	0.04	0.06	-0.09	-0.15*	1272
Coping with school day: Willingne	ss to make an effo	rt					
Treats working materials careful	0.11	0.29***	-0.04	0.19**	-0.15*	-0.33***	1273
Completes tasks with care	0.04	0.24***	-0.04	0.20**	-0.08	-0.29***	1270
Gives up fast	-0.01	-0.13	-0.00	-0.12	0.01	0.13	1271
Tries hard if task are difficult	0.13*	0.33***	0.04	0.19**	-0.09	-0.28***	1263
Coping with school day: Social int	egration						
Integrated well in class	0.10	0.18**	-0.07	0.08	-0.17**	-0.25***	1275
Has many friends in class	0.16**	0.12	-0.06	-0.05	-0.23***	-0.18**	1271
Has many new friends in class	0.08	0.05	0.00	-0.03	-0.08	-0.05	1275

Table 5 - Parenting Styles and Skills - Further Results

The table depicts the effect of each parenting style on child's skills. All controls are summarized in table A1. Initial endowment of personality traits are measured using the Big Five of the interviewed parent. Outcomes are normalized to have mean 0 and standard deviation 1. Significance of the coefficients at conventional significance levels 1%, 5%, 10% are indicated by stars ***, **, *, respectively. The last column N shows the number of observations.

their working material more careful, are better integrated and have more friends in class. Whether parents raise their child with an *authoritative* or *democratic-loving* style is not related to how the child copes with everyday school life.

In summary, we find that both non-cognitive and cognitive skills are sensitive to the parenting style. The results shown in table 4 indicate that differences in non-cognitive skills are more pronounced than in cognitive skills. In general, an *authoritative* and a *democratic-loving* style are associated with similar cognitive and non-cognitive skills. Children with *authoritative* parents have a lower listening comprehension but are more patient than children with *democratic-loving* parents. In comparison, both an *authoritarian* and an *authoritarian-inconsistent* style are systematically associated with lower skills. Children with *authoritarian-inconsistent* styles. A *democratic-loving* style is associated with less patience than any other parenting style, even an *authoritarian-inconsistent* style.

7 Discussion

Recent literature established a strong link between children's skill development and parental monetary and time investments. In this paper, we focus on the role of parenting styles, a type of parental investment that has only recently become the focus of economic research. We use a novel latent class model (LDA-S, Munro & Ng 2022) to investigate which parenting styles can actually be observed in the data. The model directly incorporates a

link between the latent classes, i.e. parenting styles, and parental education, household income and household composition. We identify four parenting styles. Two styles closely resemble Baumrind's (1991) authoritative and authoritarian style. The other two are variations of these styles. We find that parenting styles are strongly associated with household income, education and whether the child is an only child. The results suggest that constraints in both time and (non-)cognitive skills of the parents play an important role in choosing a parenting style. Analyzing how the observed styles are associated with the child's (non-)cognitive skill development, we find that children raised with an authoritative or a democratic-loving style have the most favorable outcomes. Our results show how differences in parenting styles contribute to the skill gap between children from different socioeconomic environment. Parenting styles that are associated with low household income and having more than one child are associated with lower skills of the child. As much of the literature, we rely on observational data to estimate the effect of parenting styles on skills. Therefore, one has to keep in mind that our results can only be interpreted as causal under the strong assumption that we control all factors that affect both, the choice of the parenting style and the change in skills of the child between age 4 and 7.

Our paper gives important implications for future research. As parenting styles are not directly observable, we emphasize the challenge to operationalize them in future research, a point also made by Doepke & Zilibotti (2021). Our results suggest three important considerations. First, most commonly, the researcher does not know in which dimensions parenting styles differ. Our model identifies an authoritative and authoritarian parenting style in the sense of Baumrind (1971, 1991). The two other styles are variations of them which would have been overlooked if we relied solely on the classic theoretical framework. Therefore, along with theoretical models, data-driven approaches are a crucial tool for identifying parenting styles. Second, it is important to rely on a large set of different dimensions. This helps to properly describe the latent variables or classes and to fully understand the differences between them. More importantly, an extensive set of dimensions is crucial to separate parenting styles that are similar to each other. Our results show that the parenting styles may only differ regarding a few dimensions (e.g. authoritative vs. democratic-loving or authoritarian vs. authoritarian-inconsistent). Missing dimensions which are important could lead to misleading results. In our case, a lack of distinction between authoritarian from authoritarian-inconsistent parents would make authoritarian parenting appear worse than it actually is. Authoritarian-inconsistent parenting is associated with much less favorable outcomes. Third, datadriven methods which can handle many different dimensions in an interpretable way, such as LDA-S, are a crucial tool to handle a large set of measures for parent-child interaction.

Our paper also gives important directions for policy-makers. The recent literature finds that non-cognitive skills foster cognitive skills but not vice versa and that non-cognitive skills mainly develop in childhood and hardly change in adulthood (Cunha & Heckman 2007, 2008, Cunha et al. 2010). Since parenting styles are strongly associated with non-cognitive skills, our results point to parenting styles as an important driver of the skill gap between children with different background. To reduce this gap, a policy measure, which may be easy to implement, could be to promote styles that are associated with the most favorable outcomes (authoritative or democratic-loving). However, the effectiveness of parents in implementing certain parenting styles may depend on their personal characteristics. Our results suggest that both (non-)cognitive skills and time resources of parents might limit the choice of parenting style. For example, some parents will find it harder to convince their child of their own

opinion. Such parents may have difficulties to exercise an authoritative or democratic-loving style properly. Others might just not be able to give their full attention to only one child as they have more than one. Hence, they would not be able to apply time consuming styles, e.g. democratic-loving. Policy-makers could foster parents' (non-)cognitive skills which are important to raise a child or help parents to allocate their available time between children more efficiently.

References

- Alegre, A. (2011), 'Parenting styles and children's emotional intelligence: What do we know?', *The Family Journal* **19**(1), 56–62.
- Attanasio, O., Cattan, S., Fitzsimons, E., Meghir, C. & Rubio-Codina, M. (2020), 'Estimating the production function for human capital: Results from a randomized controlled trial in Colombia', *American Economic Review* **110**(1), 48–85.
- Attanasio, O. P. (2015), 'The determinants of human capital formation during the early years of life: Theory, measurement, and policies', *Journal of the European Economic Association* **13**(6), 949–997.
- Aunola, K. & Nurmi, J.-E. (2005), 'The role of parenting styles in children's problem behavior', *Child Development* **76**(6), 1144–1159.
- Aunola, K., Stattin, H. & Nurmi, J.-E. (2000), 'Parenting styles and adolescents' achievement strategies', *Journal of Adolescence* 23(2), 205–222.
- Baumrind, D. (1971), 'Current patterns of parental authority.', Developmental Psychology 4(1p2), 1.
- Baumrind, D. (1991), 'The influence of parenting style on adolescent competence and substance use', *The Journal of Early Adolescence* **11**(1), 56–95.
- Becker, G. S. (1981), A treatise on the family, Technical report, National Bureau of Economic Research.
- Becker, G. S. & Tomes, N. (1986), 'Human capital and the rise and fall of families', *Journal of Labor Economics* **4**(3, Part 2), S1–S39.
- Becker, G. S. et al. (1960), An economic analysis of fertility. Demographic and economic change in developed countries, *in* 'NBER Conference Series', Vol. 11, pp. 209–231.
- Berendes, K., Weinert, S., Zimmermann, S. & Artelt, C. (2013), 'Assessing language indicators across the lifespan within the German National Educational Panel Study (NEPS)', *Journal for Educational Research Online* 5(2), 15–49.
- Blei, D. M., Ng, A. Y. & Jordan, M. I. (2003), 'Latent dirichlet allocation', Journal of Machine Learning Research 3(Jan), 993–1022.
- Blossfeld, H.-P. & Roßbach, H.-G. (2019), 'Education as a lifelong process: The German National Educational Panel Study (NEPS)', *Edition ZfE* (2).

- Boneva, T. & Rauh, C. (2016), 'Human capital production and parental beliefs', Unpublished Manuscript, University College London .
- Burton, P., Phipps, S. & Curtis, L. (2002), 'All in the family: A simultaneous model of parenting style and child conduct', *American Economic Review* **92**(2), 368–372.
- Cobb-Clark, D. A., Salamanca, N. & Zhu, A. (2019), 'Parenting style as an investment in human development', Journal of Population Economics **32**(4), 1315–1352.
- Cunha, F., Elo, I. & Culhane, J. (2013), Eliciting maternal expectations about the technology of cognitive skill formation, Technical report, National Bureau of Economic Research.
- Cunha, F. & Heckman, J. (2007), 'The technology of skill formation', American Economic Review 97(2), 31-47.
- Cunha, F. & Heckman, J. J. (2008), 'Formulating, identifying and estimating the technology of cognitive and noncognitive skill formation', *Journal of Human Resources* **43**(4), 738–782.
- Cunha, F., Heckman, J. J., Lochner, L. & Masterov, D. V. (2006), Interpreting the evidence on life cycle skill formation, *in* 'Handbook of the Economics of Education Volume 1', Vol. 1 of *Handbook of the Economics of Education*, Elsevier, pp. 697–812.
- Cunha, F., Heckman, J. J. & Navarro, S. (2005), 'Separating uncertainty from heterogeneity in life cycle earnings', Oxford Economic Papers 57(2), 191–261.
- Cunha, F., Heckman, J. J. & Schennach, S. M. (2010), 'Estimating the technology of cognitive and noncognitive skill formation', *Econometrica* **78**(3), 883–931.
- Dizon-Ross, R. (2016), 'Parents' beliefs and children's education: Experimental evidence from Malawi', Unpublished Manuscript, University of Chicago.
- Doepke, M., Sorrenti, G. & Zilibotti, F. (2019), 'The economics of parenting', *Annual Review of Economics* 11, 55–84.
- Doepke, M. & Zilibotti, F. (2017), 'Parenting with style: Altruism and paternalism in intergenerational preference transmission', *Econometrica* **85**(5), 1331–1371.
- Doepke, M. & Zilibotti, F. (2021), 'Do rising returns to education justify "helicopter" parenting?', *IZA World of Labor*.
- Falk, A., Kosse, F., Pinger, P., Schildberg-Hörisch, H. & Deckers, T. (2021), 'Socioeconomic status and inequalities in children's IQ and economic preferences', *Journal of Political Economy* **129**(9), 2504–2545.
- Fiorini, M. & Keane, M. P. (2014), 'How the allocation of children's time affects cognitive and noncognitive development', *Journal of Labor Economics* **32**(4), 787–836.
- Francesconi, M. & Heckman, J. J. (2016), 'Child development and parental investment: Introduction', The Economic Journal 126(596), F1–F27.

- Goodman, R. (1997), 'The strengths and difficulties questionnaire: A research note', *Journal of Child Psychology* and *Psychiatry* **38**(5), 581–586.
- Hagenaars, A. J., De Vos, K., Asghar Zaidi, M. et al. (1994), 'Poverty statistics in the late 1980s: Research based on micro-data'.
- Heckman, J. J. & Mosso, S. (2014), 'The economics of human development and social mobility', *Annual Review* of *Economics* **6**, 689–733.
- Huggett, M., Ventura, G. & Yaron, A. (2011), 'Sources of lifetime inequality', *American Economic Review* **101**(7), 2923–2954.
- Keane, M. P. & Wolpin, K. I. (1997), 'The career decisions of young men', *Journal of Political Economy* **105**(3), 473–522.
- Lundberg, S., Romich, J. L. & Tsang, K. P. (2009), 'Decision-making by children', *Review of Economics of the Household* **7**(1), 1–30.
- Masud, H., Thurasamy, R. & Ahmad, M. S. (2015), 'Parenting styles and academic achievement of young adolescents: A systematic literature review', *Quality & Quantity* **49**(6), 2411–2433.
- Munro, E. & Ng, S. (2022), 'Latent dirichlet analysis of categorical survey responses', *Journal of Business & Economic Statistics* **40**(1), 256–271.
- NEPS (2020), Information on competence testing wave 7, Technical report, LIfBi.
- NEPS (2021a), Information on competence testing wave 8, Technical report, LIfBi.
- NEPS (2021*b*), 'National Educational Panel Study, Scientific Use File of Starting Cohort Newborns', *LIfBi Leibniz* Institute for Educational Trajectories.
- Weinberg, B. A. (2001), 'An incentive model of the effect of parental income on children', *Journal of Political Economy* **109**(2), 266–280.
- Wohlkinger, F., Blumenfelder, A. R., Bayer, M., Maurice, J. v., Ditton, H. & Blossfeld, H.-P. (2019), Measuring motivational concepts and personality aspects in the National Educational Panel Study, *in* 'Education as a Lifelong Process', Springer, pp. 155–169.

A Tables

Skills at age 4		Mean	Standard Deviation	Min	Max
Mathmatical literacy 0.0444 0.9813 -3.6 3.2 Listening Comprehension 49.5915 24.0283 0.0 121.0 Prosocial behavior (SDQ) 7.7081 1.4443 1.0 10.0 Problem behavior (SDQ) 1.2479 1.3684 0.0 8.0 Patience 0.7787 0.3954 0.0 6.0 Short term memory 2.3747 0.3964 0.0 6.0 Freeling down when failing (temperament) 4.5126 1.34315 0.0 6.0 Feeling down when failing (temperament) 2.9776 1.5640 0.0 6.0 Difficult to calm down (temperament) 2.9776 1.5640 0.0 6.0 Parental Sitilis and preferences 2.6283 2.00 10.0 0.0 Parental Sitilis and preferences 3.6374 0.5472 1.0 5.0 Restroversion (Big Five) 3.6374 0.5472 1.0 5.0 Openness (Big Five) 3.6374 0.5472 1.0 7.0 Qualmonic (Big Five)	Skills at age 4				
Listening Comprehension 49.5915 24.0283 0.0 121.0 Prosocial behavior (SDQ) 7.7081 1.4843 1.0 10.0 Problem behavior (SDQ) 1.2479 1.3648 0.0 8.0 Patience 0.7787 0.3954 0.0 6.0 Frustration (temperament) 4.5126 1.2958 0.0 6.0 Econcentrated (temperament) 4.6621 1.4315 0.0 6.0 Gets lost in books (temperament) 4.96621 1.4315 0.0 6.0 Difficult to calm down (temperament) 2.9776 1.5640 0.0 10.0 Parents 2.9776 1.5640 0.0 10.0 Conscientiousness (Big Five) 3.5510 0.8313 1.0 5.0 Extraversion (Big Five) 3.53510 0.8313 1.0 5.0 Agreeableness (Big Five) 3.5374 0.5472 1.7 5.0 Openness (Big Five) 3.5374 0.5472 1.0 7.0 Parents' peers: University degree 4.4034	Reasoning	0.3037	2.3476	-4.1	6.1
Prosocial behavior (SDQ) 7.7081 1.4843 1.0 10.0 Problem behavior (SDQ) 1.2479 1.3648 0.0 8.0 Patience 0.7787 0.3954 0.0 6.0 Fustration (temperament) 4.2296 1.3682 0.0 6.0 Is concentrated (temperament) 4.5126 1.2958 0.0 6.0 Feling down when failing (temperament) 4.9430 1.3063 0.0 6.0 Difficult to calm down (temperament) 2.9776 1.5640 0.0 10.0 Parental skills and preferences 5.6288 2.2663 0.0 10.0 Patience 5.6288 2.2663 0.0 10.0 Conscientiousness (Big Five) 3.7362 0.8855 1.0 5.0 Agreeableness (Big Five) 3.7362 0.8855 1.0 5.0 Parental skills 9.0064 1.5700 7.0 7.0 Parents 'ipeers: University degree 4.4034 1.5460 1.0 7.0 Parents 'ipeers: University degree	Mathmatical literacy	0.0444	0.9813	-3.6	3.2
Problem behavior (SDQ) 1.2479 1.3648 0.0 8.0 Patience 0.7787 0.3954 0.0 1.0 Short term memory 2.3747 0.9009 0.0 5.0 Frustration (temperament) 4.5126 1.3682 0.0 6.0 Sconcentrated (temperament) 4.5126 1.4315 0.0 6.0 Fueling down when failing (temperament) 4.9430 1.3063 0.0 6.0 Full of energy in the evening (temperament) 4.9430 1.3063 0.0 6.0 Parental skills and preferences 8.6654 2.0122 0.0 10.0 Conscientiousness (Big Five) 4.0705 0.6682 1.0 5.0 Agreeableness (Big Five) 3.5310 0.3311 1.0 5.0 Neuroticism (Big Five) 3.7362 0.8855 1.0 5.0 Neuroticism (Big Five) 3.7372 0.8351 1.0 7.0 Parents' peers: Migration background 3.2917 1.3385 1.0 7.0 Parents' peers: Migration b	Listening Comprehension	49.5915	24.0283	0.0	121.0
Patience 0.7787 0.3954 0.0 1.0 Short term memory 2.3747 0.9009 0.0 5.0 Frustration (temperament) 4.2296 1.3682 0.0 6.0 Is concentrated (temperament) 4.6621 1.4315 0.0 6.0 Gets los in books (temperament) 4.9430 1.3063 0.0 6.0 Patience 4.66621 1.4315 0.0 6.0 Patience 4.6664 2.0122 0.0 10.0 Patience 5.6288 2.2663 0.0 10.0 Conscientiousness (Big Five) 3.6510 0.8313 1.0 5.0 Extraversion (Big Five) 3.6374 0.5472 1.7 5.0 Perest 2.8150 0.7745 1.0 5.0 Parents' peers: University degree 4.4034 1.5460 1.0 7.0 Quality time 0.0064 1.5770 2.5 2.50 0.70 Parents' peers: Migration background 3.2227 1.4099 1.0	Prosocial behavior (SDQ)	7.7081	1.4843	1.0	10.0
Short term memory 2.3747 0.9009 0.0 5.0 Frustration (temperament) 4.2296 1.3682 0.0 6.0 Is concentrated (temperament) 4.5126 1.2958 0.0 6.0 Feeling down when failing (temperament) 4.6621 1.4315 0.0 6.0 Full of energy in the evening (temperament) 2.9776 1.5640 0.0 6.0 Parental skills and preferences 2.9776 1.5640 0.0 6.0 Parental skills and preferences 5.6288 2.2663 0.0 10.0 Conscientiousness (Big Five) 3.6374 0.5472 1.7 5.0 Agreeableness (Big Five) 3.6374 0.5472 1.7 5.0 Openness (Big Five) 3.7362 0.8855 1.0 7.0 Parents' peers: Migration background 2.9817 1.3385 1.0 7.0 Parents' peers: Migration background 2.2817 1.3385 1.0 7.0 Quality time 9.0064 1.5770 2.5 2.8.3 <td< td=""><td>Problem behavior (SDQ)</td><td>1.2479</td><td>1.3648</td><td>0.0</td><td>8.0</td></td<>	Problem behavior (SDQ)	1.2479	1.3648	0.0	8.0
Frustration (temperament) 4.2296 1.3682 0.0 6.0 Is concentrated (temperament) 4.5126 1.2958 0.0 6.0 Feeling down when failing (temperament) 4.6621 1.4315 0.0 6.0 Gets lost in books (temperament) 4.9430 1.3063 0.0 6.0 Difficult to calm down (temperament) 2.9776 1.5640 0.0 10.0 Parental skills and preferences 5.6288 2.2663 0.0 10.0 Conscientiousness (Big Five) 3.6510 0.8313 1.0 5.0 Agreeableness (Big Five) 3.6374 0.5472 1.7 5.0 Agreeablenes (Big Five) 3.6374 0.5472 1.0 5.0 Parents (Big Five) 3.6374 0.5472 1.0 7.0 Quality time 2.9817 1.3385 1.0 7.0 Quality time 9.0064 1.5770 2.5 2.8.12 Quality time 9.0064 1.5770 2.5 2.5.1 Demographic characteristics <td< td=""><td>Patience</td><td>0.7787</td><td>0.3954</td><td>0.0</td><td>1.0</td></td<>	Patience	0.7787	0.3954	0.0	1.0
Is concentrated (temperament) 4.5126 1.2958 0.0 6.0 Feeling down when failing (temperament) 3.0783 1.5319 0.0 6.0 Gets lost in books (temperament) 4.9430 1.3063 0.0 6.0 Difficult to calm down (temperament) 2.9776 1.5640 0.0 6.0 Parental skills and preferences .	Short term memory	2.3747	0.9009	0.0	5.0
Feeling down when failing (temperament) 3.0783 1.5319 0.0 6.0 Gets lost in books (temperament) 4.9621 1.4315 0.0 6.0 Full of energy in the evening (temperament) 2.9776 1.5640 0.0 6.0 Parental Skills and preferences 5.6288 2.0122 0.0 10.0 Patience 5.6288 2.2663 0.0 10.0 Conscientiousness (Big Five) 3.6510 0.8313 1.0 5.0 Extraversion (Big Five) 3.6574 0.5472 1.7 5.0 Openness (Big Five) 3.7362 0.8855 1.0 5.0 Neuroticism (Big Five) 3.7362 0.8855 1.0 7.0 Parents' peers: Migration background 2.9817 1.3385 1.0 7.0 Quality time 9.0064 1.5770 2.5 28.3 Quality time 9.0064 1.5770 2.5 2.83 Quality time 9.0064 0.00 1.0 1.0 Child female 0.4974 <td< td=""><td>Frustration (temperament)</td><td>4.2296</td><td>1.3682</td><td>0.0</td><td>6.0</td></td<>	Frustration (temperament)	4.2296	1.3682	0.0	6.0
Gets lost in books (temperament) 4.6621 1.4315 0.0 6.0 Full of energy in the evening (temperament) 2.9776 1.5660 0.0 6.0 Parental skills and preferences 2.9776 1.5660 0.0 10.0 Patience 5.6288 2.2663 0.0 10.0 Conscientiousness (Big Five) 4.0765 0.6682 1.0 5.0 Extraversion (Big Five) 3.6374 0.5472 1.7 5.0 Openness (Big Five) 3.6374 0.5472 1.0 5.0 Parents (Big Five) 2.8150 0.7745 1.0 5.0 Parents peers: University degree 4.4034 1.5460 1.0 7.0 Quality time 2.9004 1.5770 2.5 28.3 Demographic characteristics 1.00 7.0 Child migration background 3.2237 1.4099 1.0 7.0 Quality time 0.4074 0.5002 0.0 1.0 Child migration background 0.1262 0.3321	Is concentrated (temperament)	4.5126	1.2958	0.0	6.0
Full of energy in the evening (temperament) 4.9430 1.3063 0.0 6.0 Difficult to calm down (temperament) 2.9776 1.5640 0.0 6.0 Parental skills and preferences 4.6654 2.0122 0.0 10.0 Patience 5.6288 2.2663 0.0 10.0 Conscientiousness (Big Five) 3.6510 0.8313 1.0 5.0 Extraversion (Big Five) 3.6374 0.5472 1.7 5.0 Openness (Big Five) 3.6374 0.5472 1.0 5.0 Neuroticism (Big Five) 2.8150 0.7745 1.0 5.0 Parents' peers: Migration background 2.9817 1.3385 1.0 7.0 Quality time 9.0064 1.5770 2.5 28.3 Demographic characteristics 1.0 7.0 Child female 0.4662 0.4990 0.0 1.0 Child female 0.4662 0.4990 0.0 1.0 State of Residence: HH 0.0647 0.2461 </td <td>Feeling down when failing (temperament)</td> <td>3.0783</td> <td>1.5319</td> <td>0.0</td> <td>6.0</td>	Feeling down when failing (temperament)	3.0783	1.5319	0.0	6.0
Difficult to calm down (temperament) 2.9776 1.5640 0.0 6.0 Parental skills and preferences .	Gets lost in books (temperament)	4.6621	1.4315	0.0	6.0
Parental skills and preferences Risk tolerance 4.6654 2.0122 0.0 10.0 Patience 5.6288 2.2663 0.0 10.0 Conscientiousness (Big Five) 3.6510 0.8313 1.0 5.0 Extraversion (Big Five) 3.6374 0.5472 1.7 5.0 Openness (Big Five) 3.7362 0.8855 1.0 5.0 Neuroticism (Big Five) 2.8150 0.7745 1.0 5.0 Parents 'peers: University degree 4.4034 1.5460 1.0 7.0 Parents 'peers: Migration background 3.2237 1.4099 1.0 7.0 Quality time 9.0064 1.5770 2.5 28.3 Demographic characteristics Child migration background 0.1262 0.3321 0.0 1.0 Child male 0.4974 0.5002 0.0 1.0 State of Residence: SH 0.0433 0.2047 0.0 1.0 State of Residence: SH 0.0433 0.2047 0.0 1.0 State of Residence:	Full of energy in the evening (temperament)	4.9430	1.3063	0.0	6.0
Risk tolerance 4.6654 2.0122 0.0 10.0 Patience 5.6288 2.2663 0.0 10.0 Conscientiousness (Big Five) 4.0765 0.6682 1.0 5.0 Extraversion (Big Five) 3.5510 0.8313 1.0 5.0 Agreeableness (Big Five) 3.7362 0.8855 1.0 5.0 Neuroticism (Big Five) 2.7362 0.8855 1.0 5.0 Parents' peers: University degree 4.4034 1.5460 1.0 7.0 Parents' peers: Migration background 2.9817 1.3385 1.0 7.0 Quality time 9.0064 1.5770 2.5 28.3 Demographic characteristics 2.00 1.0 1.0 Child female 0.4974 0.5002 0.0 1.0 Child migration background 0.1262 0.3321 0.0 1.0 State of Residence: SH 0.0438 0.2047 0.0 1.0 State of Residence: HH 0.0647 0.2637 0.0 <td< td=""><td>Difficult to calm down (temperament)</td><td>2.9776</td><td>1.5640</td><td>0.0</td><td>6.0</td></td<>	Difficult to calm down (temperament)	2.9776	1.5640	0.0	6.0
Patience 5.6288 2.2663 0.0 1.0.0 Conscientiousness (Big Five) 4.0765 0.6682 1.0 5.0 Extraversion (Big Five) 3.5510 0.8313 1.0 5.0 Agreeableness (Big Five) 3.6374 0.5472 1.7 5.0 Openness (Big Five) 3.7362 0.8855 1.0 5.0 Neuroticism (Big Five) 2.8150 0.7745 1.0 5.0 Parents' peers: University degree 4.4034 1.5460 1.0 7.0 Parents' peers: Migration background 3.2237 1.4099 1.0 7.0 Quality time 9.0064 1.5770 2.5 28.3 Demographic characteristics 0.10 1.0 7.0 Child female 0.4974 0.5002 0.0 1.0 Child female 0.4662 0.4990 0.0 1.0 State of Residence: SH 0.0438 0.2047 0.0 1.0 State of Residence: SH 0.0221 0.1475 0.0 1.0	Parental skills and preferences				
Conscientiousness (Big Five) 4.0765 0.6682 1.0 5.0 Extraversion (Big Five) 3.5510 0.8313 1.0 5.0 Agreeableness (Big Five) 3.6374 0.5472 1.7 5.0 Openness (Big Five) 2.8150 0.7475 1.0 5.0 Neuroticism (Big Five) 2.8150 0.745 1.0 5.0 Peress	Risk tolerance	4.6654	2.0122	0.0	10.0
Extraversion (Big Five) 3.5510 0.8313 1.0 5.0 Agreeableness (Big Five) 3.6374 0.5472 1.7 5.0 Openness (Big Five) 2.8150 0.7745 1.0 5.0 Neuroticism (Big Five) 2.8150 0.7745 1.0 5.0 Parents' peers: University degree 4.4034 1.5460 1.0 7.0 Parents' peers: Migration background 2.9817 1.3385 1.0 7.0 Quality time 9.0064 1.5770 2.5 28.3 Demographic characteristics 0.10 7.0 Child female 0.4974 0.5002 0.0 1.0 Child female 0.4974 0.5002 0.0 1.0 State of Residence: SH 0.0438 0.2047 0.0 1.0 State of Residence: HH 0.0647 0.2461 0.0 1.0 State of Residence: NW 0.2301 0.4210 0.0 1.0 State of Residence: HE 0.0569 0.2317 0.0 1.0 </td <td>Patience</td> <td>5.6288</td> <td>2.2663</td> <td>0.0</td> <td>10.0</td>	Patience	5.6288	2.2663	0.0	10.0
Extraversion (Big Five) 3.5510 0.8313 1.0 5.0 Agreeableness (Big Five) 3.6374 0.5472 1.7 5.0 Openness (Big Five) 2.8150 0.7745 1.0 5.0 Neuroticism (Big Five) 2.8150 0.7745 1.0 5.0 Parents' peers: University degree 4.4034 1.5460 1.0 7.0 Parents' peers: Migration background 2.9817 1.3385 1.0 7.0 Quality time 9.0064 1.5770 2.5 28.3 Demographic characteristics 0.10 7.0 Child female 0.4974 0.5002 0.0 1.0 Child female 0.4974 0.5002 0.0 1.0 State of Residence: SH 0.0438 0.2047 0.0 1.0 State of Residence: HH 0.0647 0.2461 0.0 1.0 State of Residence: NW 0.2301 0.4210 0.0 1.0 State of Residence: HE 0.0569 0.2317 0.0 1.0 </td <td>Conscientiousness (Big Five)</td> <td>4.0765</td> <td>0.6682</td> <td>1.0</td> <td>5.0</td>	Conscientiousness (Big Five)	4.0765	0.6682	1.0	5.0
Agreeableness (Big Five) 3.6374 0.5472 1.7 5.0 Openness (Big Five) 3.7362 0.8855 1.0 5.0 Neuroticism (Big Five) 2.8150 0.7745 1.0 5.0 Parents' peers: University degree 4.4034 1.5460 1.0 7.0 Parents' peers: Migration background 2.9817 1.3385 1.0 7.0 Quality time 9.0064 1.5770 2.5 28.3 Demographic characteristics 0.4974 0.5002 0.0 1.0 Child' speers: SH 0.4974 0.5002 0.0 1.0 Child female 0.4974 0.5002 0.0 1.0 Child female 0.4974 0.5002 0.0 1.0 Child speare: SH 0.0438 0.2047 0.0 1.0 State of Residence: SH 0.0647 0.2461 0.0 1.0 State of Residence: HB 0.0222 0.1475 0.0 1.0 State of Residence: NN 0.2301 0.4210 0.0 1.0 State of Residence: RP 0.0281 0.1653 <t< td=""><td>Extraversion (Big Five)</td><td>3.5510</td><td>0.8313</td><td>1.0</td><td>5.0</td></t<>	Extraversion (Big Five)	3.5510	0.8313	1.0	5.0
Neuroticism (Big Five) 2.8150 0.7745 1.0 5.0 Peers Parents' peers: Migration background 2.9817 1.3385 1.0 7.0 Parents' peers: Migration background 3.2237 1.4099 1.0 7.0 Quality time 9.0064 1.5770 2.5 28.3 Demographic characteristics 0.4974 0.5002 0.0 1.0 Child female 0.4974 0.5002 0.0 1.0 Child migration background 0.1262 0.3321 0.0 1.0 Responded female 0.4662 0.4990 0.0 1.0 State of Residence: SH 0.0438 0.2047 0.0 1.0 State of Residence: NH 0.0647 0.2461 0.0 1.0 State of Residence: NW 0.2012 0.1475 0.0 1.0 State of Residence: NW 0.2301 0.4210 0.0 1.0 State of Residence: RP 0.0281 0.1653 0.0 1.0 State of Residence: BW	(=)	3.6374	0.5472	1.7	5.0
Neuroticism (Big Five) 2.8150 0.7745 1.0 5.0 Peers Parents' peers: Migration background 2.9817 1.3385 1.0 7.0 Parents' peers: Migration background 3.2237 1.4099 1.0 7.0 Quality time 9.0064 1.5770 2.5 28.3 Demographic characteristics 0.4974 0.5002 0.0 1.0 Child female 0.4974 0.5002 0.0 1.0 Child migration background 0.1262 0.3321 0.0 1.0 Responded female 0.4662 0.4990 0.0 1.0 State of Residence: SH 0.0438 0.2047 0.0 1.0 State of Residence: NH 0.0647 0.2461 0.0 1.0 State of Residence: NW 0.2012 0.1475 0.0 1.0 State of Residence: NW 0.2301 0.4210 0.0 1.0 State of Residence: RP 0.0281 0.1653 0.0 1.0 State of Residence: BW				1.0	5.0
Peers Parents' peers: University degree 4.4034 1.5460 1.0 7.0 Parents' peers: Migration background 2.9817 1.3385 1.0 7.0 Child's peers: Migration background 3.2237 1.4099 1.0 7.0 Quality time 9.0064 1.5770 2.5 28.3 Demographic characteristics 0.4974 0.5002 0.0 1.0 Child female 0.4974 0.5002 0.0 1.0 1.0 Responded female 0.4662 0.4990 0.0 1.0 State of Residence: SH 0.0438 0.2047 0.0 1.0 State of Residence: HH 0.0647 0.2461 0.0 1.0 State of Residence: NI 0.0752 0.2637 0.0 1.0 State of Residence: NW 0.2301 0.4210 0.0 1.0 State of Residence: HB 0.0222 0.1475 0.0 1.0 State of Residence: BW 0.0915 0.2884 0.0 1.0		2.8150		1.0	
Parents' peers: Migration background 2.9817 1.3385 1.0 7.0 Quality time 3.2237 1.4099 1.0 7.0 Quality time 9.0064 1.5770 2.5 28.3 Demographic characteristics 0.4974 0.5002 0.0 1.0 Child female 0.4974 0.5002 0.0 1.0 Child female 0.4662 0.4990 0.0 1.0 State of Residence: SH 0.0438 0.2047 0.0 1.0 State of Residence: HH 0.0647 0.2461 0.0 1.0 State of Residence: NI 0.0752 0.2637 0.0 1.0 State of Residence: NW 0.2301 0.4210 0.0 1.0 State of Residence: HE 0.0569 0.2317 0.0 1.0 State of Residence: RP 0.0281 0.1633 0.0 1.0 State of Residence: BW 0.0137 0.1164 0.0 1.0 State of Residence: BE 0.1078 0.3103 0.0 <	(= /			-	
Parents' peers: Migration background 2.9817 1.3385 1.0 7.0 Quality time 3.2237 1.4099 1.0 7.0 Quality time 9.0064 1.5770 2.5 28.3 Demographic characteristics 0.4974 0.5002 0.0 1.0 Child female 0.4974 0.5002 0.0 1.0 Child female 0.4662 0.4990 0.0 1.0 State of Residence: SH 0.0438 0.2047 0.0 1.0 State of Residence: HH 0.0647 0.2461 0.0 1.0 State of Residence: NI 0.0752 0.2637 0.0 1.0 State of Residence: NW 0.2301 0.4210 0.0 1.0 State of Residence: HE 0.0569 0.2317 0.0 1.0 State of Residence: RP 0.0281 0.1633 0.0 1.0 State of Residence: BW 0.0137 0.1164 0.0 1.0 State of Residence: BE 0.1078 0.3103 0.0 <	Parents' neers: University degree	4 4034	1 5460	1.0	7.0
Child's peers: Migration background 3.2237 1.4099 1.0 7.0 Quality time 9.0064 1.5770 2.5 28.3 Demographic characteristics 0.5002 0.0 1.0 Child female 0.4974 0.5002 0.0 1.0 Child female 0.4662 0.4990 0.0 1.0 Responded female 0.4662 0.4990 0.0 1.0 State of Residence: SH 0.0438 0.2047 0.0 1.0 State of Residence: HH 0.0647 0.2461 0.0 1.0 State of Residence: NI 0.0752 0.2637 0.0 1.0 State of Residence: HB 0.0222 0.1475 0.0 1.0 State of Residence: HE 0.0569 0.2317 0.0 1.0 State of Residence: RP 0.0281 0.1653 0.0 1.0 State of Residence: BW 0.0137 0.1164 0.0 1.0 State of Residence: BE 0.1078 0.3103 0.0 1.0					
Quality time 9.0064 1.5770 2.5 28.3 Demographic characteristics 2 2 2 2 3 Child female 0.4974 0.5002 0.0 1.0 Child migration background 0.1262 0.3321 0.0 1.0 Responded female 0.4662 0.4990 0.0 1.0 State of Residence: SH 0.0438 0.2047 0.0 1.0 State of Residence: HH 0.0647 0.2461 0.0 1.0 State of Residence: NI 0.0752 0.2637 0.0 1.0 State of Residence: HB 0.0222 0.1475 0.0 1.0 State of Residence: NW 0.2301 0.4210 0.0 1.0 State of Residence: HE 0.0569 0.2317 0.0 1.0 State of Residence: BW 0.0915 0.2884 0.0 1.0 State of Residence: SI 0.0137 0.1164 0.0 1.0 State of Residence: SI 0.0178 0.3103 0.0					
Quality time 9.0064 1.5770 2.5 28.3 Demographic characteristics		0.2201	1.1000	2.0	
Demographic characteristics Child female 0.4974 0.5002 0.0 1.0 Child migration background 0.1262 0.3321 0.0 1.0 Responded female 0.4662 0.4990 0.0 1.0 State of Residence: SH 0.0438 0.2047 0.0 1.0 State of Residence: HH 0.0647 0.2461 0.0 1.0 State of Residence: NI 0.0752 0.2637 0.0 1.0 State of Residence: HB 0.0222 0.1475 0.0 1.0 State of Residence: HE 0.0569 0.2317 0.0 1.0 State of Residence: RP 0.0281 0.1653 0.0 1.0 State of Residence: BW 0.0915 0.2884 0.0 1.0 State of Residence: SL 0.0137 0.1164 0.0 1.0 State of Residence: SL 0.0137 0.1164 0.0 1.0 State of Residence: BB 0.0124 0.1108 0.0 1.0 State of Residence: SN		9 0064	1 5770	2.5	28.3
Child female 0.4974 0.5002 0.0 1.0 Child migration background 0.1262 0.3321 0.0 1.0 Responded female 0.4662 0.4990 0.0 1.0 State of Residence: SH 0.0438 0.2047 0.0 1.0 State of Residence: HH 0.0647 0.2461 0.0 1.0 State of Residence: NI 0.0752 0.2637 0.0 1.0 State of Residence: HB 0.0222 0.1475 0.0 1.0 State of Residence: HB 0.0222 0.1475 0.0 1.0 State of Residence: HE 0.0569 0.2317 0.0 1.0 State of Residence: RP 0.0281 0.1653 0.0 1.0 State of Residence: BY 0.1641 0.3704 0.0 1.0 State of Residence: SL 0.0137 0.1164 0.0 1.0 State of Residence: BE 0.1078 0.3103 0.0 1.0 State of Residence: BB 0.0124 0.1108 0.0		5.0001	1.0110	2.0	2010
Child migration background0.12620.33210.01.0Responded female0.46620.49900.01.0State of Residence: SH0.04380.20470.01.0State of Residence: HH0.06470.24610.01.0State of Residence: NI0.07520.26370.01.0State of Residence: HB0.02220.14750.01.0State of Residence: NW0.23010.42100.01.0State of Residence: HE0.05690.23170.01.0State of Residence: RP0.02810.16530.01.0State of Residence: BW0.09150.28840.01.0State of Residence: BY0.16410.37040.01.0State of Residence: BE0.10780.31030.01.0State of Residence: BE0.10780.31030.01.0State of Residence: BB0.01240.11080.01.0State of Residence: SN0.04250.20180.01.0State of Residence: ST0.03330.17960.01.0State of Residence: TH0.00720.08450.01.0State of Residence: TH0.03330.17960.01.0Group: No sibling, low income, no univ.0.03330.17960.01.0Group: No sibling, low income, no univ.0.04770.21320.01.0Group: No sibling, middle income., no univ.0.04770.21320.01.0Gro	• •	0 4074	0.5002	0.0	1.0
Responded female 0.4662 0.4990 0.0 1.0 State of Residence: SH 0.0438 0.2047 0.0 1.0 State of Residence: HH 0.0647 0.2461 0.0 1.0 State of Residence: HH 0.0752 0.2637 0.0 1.0 State of Residence: HB 0.0222 0.1475 0.0 1.0 State of Residence: HB 0.02301 0.4210 0.0 1.0 State of Residence: HE 0.0569 0.2317 0.0 1.0 State of Residence: RP 0.0281 0.1653 0.0 1.0 State of Residence: BW 0.0915 0.2884 0.0 1.0 State of Residence: SL 0.0137 0.1164 0.0 1.0 State of Residence: BE 0.1078 0.3103 0.0 1.0 State of Residence: BB 0.0124 0.1108 0.0 1.0 State of Residence: SN 0.0425 0.2018 0.0 1.0 State of Residence: ST 0.0235 0.1516 0.0					
State of Residence: SH 0.0438 0.2047 0.0 1.0 State of Residence: HH 0.0647 0.2461 0.0 1.0 State of Residence: NI 0.0752 0.2637 0.0 1.0 State of Residence: HB 0.0222 0.1475 0.0 1.0 State of Residence: HB 0.0222 0.1475 0.0 1.0 State of Residence: HE 0.0569 0.2317 0.0 1.0 State of Residence: RP 0.0281 0.1653 0.0 1.0 State of Residence: BW 0.0915 0.2884 0.0 1.0 State of Residence: BY 0.1641 0.3704 0.0 1.0 State of Residence: SL 0.0137 0.1164 0.0 1.0 State of Residence: BE 0.1078 0.3103 0.0 1.0 State of Residence: MV 0.0163 0.1268 0.0 1.0 State of Residence: SN 0.0425 0.2018 0.0 1.0 State of Residence: ST 0.0333 0.1796 0.0 1.0 State of Residence: TH 0.0333 0.1796 <td></td> <td></td> <td></td> <td></td> <td></td>					
State of Residence: HH 0.0647 0.2461 0.0 1.0 State of Residence: NI 0.0752 0.2637 0.0 1.0 State of Residence: HB 0.0222 0.1475 0.0 1.0 State of Residence: NW 0.2301 0.4210 0.0 1.0 State of Residence: HE 0.0569 0.2317 0.0 1.0 State of Residence: RP 0.0281 0.1653 0.0 1.0 State of Residence: BW 0.0915 0.2884 0.0 1.0 State of Residence: BV 0.1641 0.3704 0.0 1.0 State of Residence: SL 0.0137 0.1164 0.0 1.0 State of Residence: BE 0.1078 0.3103 0.0 1.0 State of Residence: BB 0.0124 0.1108 0.0 1.0 State of Residence: SN 0.0425 0.2018 0.0 1.0 State of Residence: ST 0.0235 0.1516 0.0 1.0 State of Residence: TH 0.0072 0.0845 0.0 1.0 State of Residence: TH 0.0333 0.1796 <td>•</td> <td></td> <td></td> <td></td> <td></td>	•				
State of Residence: NI 0.0752 0.2637 0.0 1.0 State of Residence: HB 0.0222 0.1475 0.0 1.0 State of Residence: NW 0.2301 0.4210 0.0 1.0 State of Residence: HE 0.0569 0.2317 0.0 1.0 State of Residence: RP 0.0281 0.1653 0.0 1.0 State of Residence: BW 0.0915 0.2884 0.0 1.0 State of Residence: BV 0.1641 0.3704 0.0 1.0 State of Residence: SL 0.0137 0.1164 0.0 1.0 State of Residence: BE 0.1078 0.3103 0.0 1.0 State of Residence: BB 0.0124 0.1108 0.0 1.0 State of Residence: SN 0.0425 0.2018 0.0 1.0 State of Residence: ST 0.0333 0.1796 0.0 1.0 State of Residence: TH 0.0072 0.0845 0.0 1.0 Group: No sibling, low income, no univ. 0.0333 0.1796 0.0 1.0 Group: No sibling, wincome, univ. 0.0					
State of Residence: HB 0.0222 0.1475 0.0 1.0 State of Residence: NW 0.2301 0.4210 0.0 1.0 State of Residence: HE 0.0569 0.2317 0.0 1.0 State of Residence: RP 0.0281 0.1653 0.0 1.0 State of Residence: BW 0.0915 0.2884 0.0 1.0 State of Residence: BY 0.1641 0.3704 0.0 1.0 State of Residence: SL 0.0137 0.1164 0.0 1.0 State of Residence: BE 0.1078 0.3103 0.0 1.0 State of Residence: BB 0.0124 0.1108 0.0 1.0 State of Residence: SN 0.0425 0.2018 0.0 1.0 State of Residence: ST 0.0235 0.1516 0.0 1.0 State of Residence: TH 0.0072 0.0845 0.0 1.0 Group: No sibling, low income, no univ. 0.0333 0.1796 0.0 1.0 Group: No sibling, wincome, univ. 0.0477 0.2132 0.0 1.0 Group: No sibling, middle income., univ.<					
State of Residence: NW 0.2301 0.4210 0.0 1.0 State of Residence: HE 0.0569 0.2317 0.0 1.0 State of Residence: RP 0.0281 0.1653 0.0 1.0 State of Residence: BW 0.0915 0.2884 0.0 1.0 State of Residence: BY 0.1641 0.3704 0.0 1.0 State of Residence: SL 0.0137 0.1164 0.0 1.0 State of Residence: BE 0.1078 0.3103 0.0 1.0 State of Residence: BB 0.0124 0.1108 0.0 1.0 State of Residence: SN 0.0425 0.2018 0.0 1.0 State of Residence: ST 0.0235 0.1516 0.0 1.0 State of Residence: TH 0.0072 0.0845 0.0 1.0 State of Residence: TH 0.0333 0.1796 0.0 1.0 Group: No sibling, low income, no univ. 0.0144 0.1191 0.0 1.0 Group: No sibling, middle income., no univ. 0.0477 0.2132 0.0 1.0 Group: No sibling, middle incom					
State of Residence: HE 0.0569 0.2317 0.0 1.0 State of Residence: RP 0.0281 0.1653 0.0 1.0 State of Residence: BW 0.0915 0.2884 0.0 1.0 State of Residence: BY 0.1641 0.3704 0.0 1.0 State of Residence: SL 0.0137 0.1164 0.0 1.0 State of Residence: BE 0.1078 0.3103 0.0 1.0 State of Residence: BB 0.0124 0.1108 0.0 1.0 State of Residence: SN 0.0425 0.2018 0.0 1.0 State of Residence: ST 0.0235 0.1516 0.0 1.0 State of Residence: TH 0.0072 0.0845 0.0 1.0 Group: No sibling, low income, no univ. 0.0144 0.1191 0.0 1.0 Group: No sibling, middle income., no univ. 0.0477 0.2132 0.0 1.0 Group: No sibling, middle income., univ. 0.0477 0.2132 0.0 1.0					
State of Residence: RP 0.0281 0.1653 0.0 1.0 State of Residence: BW 0.0915 0.2884 0.0 1.0 State of Residence: BY 0.1641 0.3704 0.0 1.0 State of Residence: SL 0.0137 0.1164 0.0 1.0 State of Residence: BE 0.1078 0.3103 0.0 1.0 State of Residence: BB 0.0124 0.1108 0.0 1.0 State of Residence: SN 0.0425 0.2018 0.0 1.0 State of Residence: ST 0.0235 0.1516 0.0 1.0 State of Residence: TH 0.0072 0.0845 0.0 1.0 Group: No sibling, low income, no univ. 0.0144 0.1191 0.0 1.0 Group: No sibling, middle income., no univ. 0.0477 0.2132 0.0 1.0 Group: No sibling, middle income., univ. 0.0477 0.2132 0.0 1.0					
State of Residence: BW 0.0915 0.2884 0.0 1.0 State of Residence: BY 0.1641 0.3704 0.0 1.0 State of Residence: SL 0.0137 0.1164 0.0 1.0 State of Residence: BE 0.1078 0.3103 0.0 1.0 State of Residence: BB 0.0124 0.1108 0.0 1.0 State of Residence: SN 0.0425 0.2018 0.0 1.0 State of Residence: ST 0.0235 0.1516 0.0 1.0 State of Residence: TH 0.0072 0.0845 0.0 1.0 Group: No sibling, low income, no univ. 0.0144 0.1191 0.0 1.0 Group: No sibling, middle income., no univ. 0.0477 0.2132 0.0 1.0 Group: No sibling, middle income, univ. 0.0477 0.2132 0.0 1.0					
State of Residence: BY 0.1641 0.3704 0.0 1.0 State of Residence: SL 0.0137 0.1164 0.0 1.0 State of Residence: BE 0.1078 0.3103 0.0 1.0 State of Residence: BB 0.0124 0.1108 0.0 1.0 State of Residence: MV 0.0163 0.1268 0.0 1.0 State of Residence: SN 0.0425 0.2018 0.0 1.0 State of Residence: ST 0.0235 0.1516 0.0 1.0 State of Residence: TH 0.0072 0.0845 0.0 1.0 Group: No sibling, low income, no univ. 0.0333 0.1796 0.0 1.0 Group: No sibling, middle income., no univ. 0.0477 0.2132 0.0 1.0 Group: No sibling, middle income, univ. 0.0477 0.2132 0.0 1.0					
State of Residence: SL 0.0137 0.1164 0.0 1.0 State of Residence: BE 0.1078 0.3103 0.0 1.0 State of Residence: BB 0.0124 0.1108 0.0 1.0 State of Residence: MV 0.0163 0.1268 0.0 1.0 State of Residence: SN 0.0425 0.2018 0.0 1.0 State of Residence: ST 0.0235 0.1516 0.0 1.0 State of Residence: TH 0.0072 0.0845 0.0 1.0 Group: No sibling, low income, no univ. 0.0144 0.1191 0.0 1.0 Group: No sibling, middle income., no univ. 0.0477 0.2132 0.0 1.0 Group: No sibling, middle income, univ. 0.0242 0.1537 0.0 1.0					
State of Residence: BE 0.1078 0.3103 0.0 1.0 State of Residence: BB 0.0124 0.1108 0.0 1.0 State of Residence: MV 0.0163 0.1268 0.0 1.0 State of Residence: SN 0.0425 0.2018 0.0 1.0 State of Residence: ST 0.0235 0.1516 0.0 1.0 State of Residence: TH 0.0072 0.0845 0.0 1.0 Group: No sibling, low income, no univ. 0.0333 0.1796 0.0 1.0 Group: No sibling, niddle income., no univ. 0.0477 0.2132 0.0 1.0 Group: No sibling, middle income, univ. 0.0242 0.1537 0.0 1.0					
State of Residence: BB 0.0124 0.1108 0.0 1.0 State of Residence: MV 0.0163 0.1268 0.0 1.0 State of Residence: SN 0.0425 0.2018 0.0 1.0 State of Residence: ST 0.0235 0.1516 0.0 1.0 State of Residence: TH 0.0072 0.0845 0.0 1.0 Group: No sibling, low income, no univ. 0.0333 0.1796 0.0 1.0 Group: No sibling, low income, univ. 0.0444 0.1191 0.0 1.0 Group: No sibling, middle income., no univ. 0.0477 0.2132 0.0 1.0 Group: No sibling, middle income, univ. 0.0242 0.1537 0.0 1.0					
State of Residence: MV 0.0163 0.1268 0.0 1.0 State of Residence: SN 0.0425 0.2018 0.0 1.0 State of Residence: ST 0.0235 0.1516 0.0 1.0 State of Residence: TH 0.0072 0.0845 0.0 1.0 Group: No sibling, low income, no univ. 0.0333 0.1796 0.0 1.0 Group: No sibling, low income, univ. 0.0144 0.1191 0.0 1.0 Group: No sibling, middle income., no univ. 0.0477 0.2132 0.0 1.0 Group: No sibling, middle income, univ. 0.0242 0.1537 0.0 1.0					1.0
State of Residence: SN 0.0425 0.2018 0.0 1.0 State of Residence: ST 0.0235 0.1516 0.0 1.0 State of Residence: TH 0.0072 0.0845 0.0 1.0 Group: No sibling, low income, no univ. 0.0333 0.1796 0.0 1.0 Group: No sibling, low income, univ. 0.0144 0.1191 0.0 1.0 Group: No sibling, middle income., no univ. 0.0477 0.2132 0.0 1.0 Group: No sibling, middle income, univ. 0.0242 0.1537 0.0 1.0					
State of Residence: ST 0.0235 0.1516 0.0 1.0 State of Residence: TH 0.0072 0.0845 0.0 1.0 Group: No sibling, low income, no univ. 0.0333 0.1796 0.0 1.0 Group: No sibling, low income, univ. 0.0144 0.1191 0.0 1.0 Group: No sibling, middle income., no univ. 0.0477 0.2132 0.0 1.0 Group: No sibling, middle income, univ. 0.0242 0.1537 0.0 1.0					
State of Residence: TH 0.0072 0.0845 0.0 1.0 Group: No sibling, low income, no univ. 0.0333 0.1796 0.0 1.0 Group: No sibling, low income, univ. 0.0144 0.1191 0.0 1.0 Group: No sibling, middle income., no univ. 0.0477 0.2132 0.0 1.0 Group: No sibling, middle income, univ. 0.0242 0.1537 0.0 1.0					
Group: No sibling, low income, no univ. 0.0333 0.1796 0.0 1.0 Group: No sibling, low income, univ. 0.0144 0.1191 0.0 1.0 Group: No sibling, middle income., no univ. 0.0477 0.2132 0.0 1.0 Group: No sibling, middle income, univ. 0.0242 0.1537 0.0 1.0					1.0
Group: No sibling, low income, univ. 0.0144 0.1191 0.0 1.0 Group: No sibling, middle income, no univ. 0.0477 0.2132 0.0 1.0 Group: No sibling, middle income, univ. 0.0242 0.1537 0.0 1.0		0.0072	0.0845	0.0	1.0
Group: No sibling, middle income., no univ. 0.0477 0.2132 0.0 1.0 Group: No sibling, middle income, univ. 0.0242 0.1537 0.0 1.0	Group: No sibling, low income, no univ.	0.0333	0.1796	0.0	1.0
Group: No sibling, middle income, univ. 0.0242 0.1537 0.0 1.0	Group: No sibling, low income, univ.	0.0144	0.1191	0.0	1.0
	Group: No sibling, middle income., no univ.	0.0477	0.2132	0.0	1.0
Group: No sibling, high income, no univ. 0.0255 0.1577 0.0 1.0	Group: No sibling, middle income, univ.	0.0242	0.1537	0.0	1.0
	Group: No sibling, high income, no univ.	0.0255	0.1577	0.0	1.0

Table A1 continued from previous page

Group: No sibling, high income, univ.	0.0660	0.2484	0.0	1.0
Group: Sibling, low income, no univ.	0.1614	0.3681	0.0	1.0
Group: Sibling., low income, univ.	0.0725	0.2595	0.0	1.0
Group: Sibling., middle income, no univ.	0.0915	0.2884	0.0	1.0
Group: Sibling., middle income, univ.	0.1660	0.3722	0.0	1.0
Group: Sibling., high income, no univ.	0.0608	0.2390	0.0	1.0
Group: Sibling, high income, univ.	0.2366	0.4251	0.0	1.0
Ν	1530			

Table A1 – Summary statistics

IAW-Diskussionspapiere

Die IAW-Diskussionspapiere erscheinen seit September 2001. Die vollständige Liste der IAW-Diskussionspapiere von 2001 bis 2013 (Nr. 1-110) finden Sie auf der IAW-Internetseite http://www.iaw.edu/index.php/IAW-Diskussionspapiere.

IAW-Diskussionspapiere ab 2013:

Nr. 111 Offshoring Potential and Employment Dynamics Bernhard Boockmann	(Oktober 2014)
Nr. 112 Is Offshoring Linked to Offshoring Potentials? Evidence from German Linked-Employer-Employee Data Tobias Brändle	(Oktober 2014)
Nr. 113 University Knowledge and Firm Innovation – Evidence from European Countries Andrea Bellucci / Luca Pennacchio	(November 2014)
Nr. 114 We Want them all Covered! Collective Bargaining and Firm Heterogeneity. Theory and Evidence from Germany Florian Baumann / Tobias Brändle	(Januar 2015)
Nr. 115 Coaching, Counseling, Case-Working: Do They Help the Older Unemployed out of Benefit Receipt and back into the Labor Market? Bernhard Boockmann / Tobias Brändle	(Januar 2015)
Nr. 116 The One Constant: A Causal Effect of Collective Bargaining on Employment Growth? Evidence from German Linked-Employer-Employee Data Tobias Brändle / Laszlo Goerke	(Januar 2015)
Nr. 117 Activation as a First Step: Estimating the Causal Effects of a Job Search Assistance Programme Tobias Brändle / Lukas Fervers / Carina Webel	(Februar 2015)
Nr. 118 Mental Representation of Sharing Experiments: Analyzing Choice and Belief Data Werner Güth /Charlotte Klempt / Kerstin Pull	(März 2015)
Nr. 119 Collateral and Local Lending: Testing the Lender-Based Theory Andrea Bellucci / Alexander Borisov / Germana Giombini / Alberto Zazzaro	(April 2015)
Nr. 120 The Great Trade Collapse and the Spanish Export Miracle: Firm-level Evidence from the Crisis Peter S. Eppinger / Nicole Meythaler / Marc-Manuel Sindlinger / Marcel Smolka	(April 2015)
Nr. 121 Do exporting firms benefit from retail internationalization? Angela Cheptea / Charlotte Emlinger / Karine Latouche	(Juni 2015)
Nr. 122 Bank Organization and Loan Contracting in Small Business Financing Andrea Bellucci / Alexander Borisov/ Alberto Zazzaro	(Januar 2016)
Nr. 123 Mentoring Disadvantaged Youths during School-to-Work Transition: Evidence from Germany Bernhard Boockmann / Sebastian Nielen	(März 2016)

IAW-Diskussionspapiere

Nr. 124 Ökonomische Wirkungen der Handwerksnovelle 2004: Ergebnisse aus einem quasinatürlichen Experiment Andreas Koch / Sebastian Nielen	(Mai 2016)
Nr. 125 Fast track to the labour market or highway to hell? The effect of activation policies on quantity and quality of labour market integration Lukas Fervers	(Juni 2016)
Nr. 126 Handelsliberalisierung: Unilateral, regional, multilateral Wilhelm Kohler	(Oktober 2016)
Nr. 127 Trade creating oder Trade diverting – Ökonomische Perspektiven auf den Abbau technischer Handelshemmnisse in multilateralem oder regionalem Rahmen Benjamin Jung	(Oktober 2016)
Nr. 128 The Distribution of Household Savings in Germany Jochen Späth / Kai Daniel Schmid	(November 2016)
Nr. 129 Give it Another Try: What are the Effects of a Public Employment Scheme Especially Designed for Hard-to-Place Workers? Tobias Brändle / Lukas Fervers	(April 2017)
Nr. 130 Integration der Geflüchteten in den Arbeitsmarkt: Ziele, Strategien und Hemmnisse aus Sicht der Jobcenter Bernhard Boockmann / Tobias Scheu	(Januar 2018)
Nr. 131 Operationalizing Seasonal Work in Germany Jochen Späth / Tobias Brändle / Stefan Preuss / Marcel Reiner	(September 2018)
Operationalizing Seasonal Work in Germany	(September 2018) (Oktober 2018)
Operationalizing Seasonal Work in Germany Jochen Späth / Tobias Brändle / Stefan Preuss / Marcel Reiner Nr. 132 Rather a trigger than a cause of change. Responses of firms and workers to the statutory minimum wage in Germany	
Operationalizing Seasonal Work in Germany Jochen Späth / Tobias Brändle / Stefan Preuss / Marcel Reiner Nr. 132 Rather a trigger than a cause of change. Responses of firms and workers to the statutory minimum wage in Germany Andreas Koch / Andrea Kirchmann / Marcel Reiner / Tobias Scheu / Holger Bonin Nr. 133 Accounting for Investment Risk in Educational Decisions: New Evidence for Lifetime Returns in Germany	(Oktober 2018)
Operationalizing Seasonal Work in Germany Jochen Späth / Tobias Brändle / Stefan Preuss / Marcel Reiner Nr. 132 Rather a trigger than a cause of change. Responses of firms and workers to the statutory minimum wage in Germany Andreas Koch / Andrea Kirchmann / Marcel Reiner / Tobias Scheu / Holger Bonin Nr. 133 Accounting for Investment Risk in Educational Decisions: New Evidence for Lifetime Returns in Germany Anne Zühlke / Philipp Kugler / Armin Hackenberger / Tobias Brändle Nr. 134 The Attractiveness of Jobs in the German Care Sector – Results of a Factorial Survey	(Oktober 2018) (April 2020)
Operationalizing Seasonal Work in Germany Jochen Späth / Tobias Brändle / Stefan Preuss / Marcel Reiner Nr. 132 Rather a trigger than a cause of change. Responses of firms and workers to the statutory minimum wage in Germany Andreas Koch / Andrea Kirchmann / Marcel Reiner / Tobias Scheu / Holger Bonin Nr. 133 Accounting for Investment Risk in Educational Decisions: New Evidence for Lifetime Returns in Germany Anne Zühlke / Philipp Kugler / Armin Hackenberger / Tobias Brändle Nr. 134 The Attractiveness of Jobs in the German Care Sector – Results of a Factorial Survey Martin Kroczek / Jochen Späth Nr. 135 Wage expectation, information and the decision to become a nurse	(Oktober 2018) (April 2020) (Dezember 2020)
Operationalizing Seasonal Work in Germany Jochen Späth / Tobias Brändle / Stefan Preuss / Marcel Reiner Nr. 132 Rather a trigger than a cause of change. Responses of firms and workers to the statutory minimum wage in Germany Andreas Koch / Andrea Kirchmann / Marcel Reiner / Tobias Scheu / Holger Bonin Nr. 133 Accounting for Investment Risk in Educational Decisions: New Evidence for Lifetime Returns in Germany Anne Zühlke / Philipp Kugler / Armin Hackenberger / Tobias Brändle Nr. 134 The Attractiveness of Jobs in the German Care Sector – Results of a Factorial Survey Martin Kroczek / Jochen Späth Nr. 135 Wage expectation, information and the decision to become a nurse Philipp Kugler Nr. 136 Analyzing Nurses' Decisions to Leave Their Profession – a Duration Analysis	(Oktober 2018) (April 2020) (Dezember 2020) (Januar 2021)